

N-96-07 II-A-918

Septembrie

ŕ

i ve

5



#### American Gas Association, Inc.

Noise Control for Reciprocating and Turbine Engines driven by Natural Gas and Liquid Fuel

December 1969

Prepared for

AMERICAN GAS ASSOCIATION, INC. 605 THIRD AVENUE NEW YORK, NEW YORK 10016

BOLT BERANEK AND NEWMAN INC. 50 Moulton Street Cambridge, Massachusetts 02138

Printed in U.S.A.

A.G.A. Catalog No. S20069

# PREFACE

This manual has been prepared for the American Gas Association and its member companies by the acoustical consulting firm of Bolt Beranek and Newman Inc. This semi-technical manual is directed toward architects, mechanical engineers, structural engineers, building owners, potential natural gas customers and gas company personnel.

-3 - .

Noise of reciprocating and turbine engines, fueled by natural gas and liquid fuel, can be controlled, and many successful installations throughout the country attest to this. This manual provides data, guidelines and procedures for the design of quiet installations.

Although the engine noise data summarized in this manual represent a rather thorough study of over 120 turbine and reciprocating engines, there is no intention to suggest that these data are so complete that they could form the basis for setting up a "standard" on engine noise. This is not the objective of this manual. In fact engine noise data are offered only to permit a quantitative approach to noise control. Please do not consider this manual as any type of industrial "standard" on engine noise!

In providing this manual to the engineering public, neither the American Gas Association nor its member companies nor Bolt Beranek and Newman Inc. assumes any legal or technical responsibility for any designs based on the use of the manual.

The author wishes to thank A.G.A. and members of its Prime Mover and Large Tonnage Air Conditioning Sales Promotion Committee and James J. Kennedy, A.G.A. Manager of Prime Mover and Large Tonnage Promotion, for their interest and assistance in the preparation of this manual.

فيتجربها العرجا الإرجابية والمتحصين والمراجا والمتحادث

LAYMON N. MILLER BOLT BERANEK AND NEWMAN INC.

1.0

-----

March 1969

armenter and the second se

S



In Caterpillar Technical Center, engines are tested in test cell beside each control room. Both rooms have liberal use of acoustical absorption material to reduce noise levels. Double-glass viewing window and separate doors from each room into common corridor further reduce noise transmission into control room. When inside test cell, engineers and operators wear ear protection. (Photograph is courtesy of Caterpillar Tractor Company.)

÷

 $\mathbf{\hat{T}}$ 

# CONTENTS

		Page
OUTLIN	E OF MANUAL	1
SECTION	N 1. THE NOISE PROBLEM	3
	t. I Noise Criteria	3
	1.2 Sound Pressure Level ("SPL")	5
	1.3 Sound Power Level ("PWL")	6
aromos		~
SECTION	v 2. THE VIBRATION PROBLEM	/
SECTION	3. RECIPROCATING ENGINE NOISE	9
	3.1 Engine Casing Noise	11
	3.2 Example, Data Sheet 1.	11
	3.3 Engine Exhaust Noise	11
	3.4 Example, Data Sheet 2.	11
	3.5 Engine Air-Intake Noise	14
	3.6 Example, Data Sheet 3.	14
	3.7 Vibration Data	14
SECTION	A TURBINE ENGINE NOISE	15
01.011011	4 1 Engine Coring Noise	15
	4.1 Engine Casing Noise	15
	4.2 Engine Air Intaka Noise	16
	4.5 Engine An Intake Noise	16
	4.5 Vibration Data	16
		10
SECTION	5. CONTROL OF AIRBORNE SOUND INDOORS	19
	5.1 Sound Pressure Level in a Room	19
	5.2 Example, Data Sheet 5	20
	5.3 Addition of Decibel Levels	20
	5.4 Transmission Loss of Walls ("TL")	20
	5.5 Noise Reduction of Walls ("NR")	22
	5.6 Example, Data Sheet 6	23
	5.7 Doors and Windows	25
	5.8 Transmission Loss of Floor-Ceiling Combinations	26
	5.9 Noise Reduction of Floor-Ceiling Combinations	28
	5.10 Example, Data Sheet 7	28
	5.11 Special Situations,	30
	5.12 Engine Room Precautions	52

nun prentra della scienza della considerata dell'activitati e della considerata considerata della considerata d

# CONTENTS

SECT	ION 6 CONTROL OF AIRBORNE SOUND OUT-OF-DOORS 33	
0401	61 Sound Program Lavel Out of Doors	
	6.7 Example Data Sheet 8	
	6.3 Noise Reduction Provided by a Building 33	
	6.4 Noise Reduction Provided by a Solid Partiet	
	6.5 Noise Reduction Provided by Dense Woods	
	6.6 Directivity Effect of a Stack Opening	
	6.7 Noise Reduction Provided by "Reactive Mufflers" 37	
	6.8 Noise Reduction Provided by "Dissipative Mufflers"	
	6.9 Example. Data Sheet 9	
	6.10 Noise Control for an Outdoor Engine	
	6.11 Noise Escape from an Opening	
	6.12 Caution on Outdoor Engine Noise	
	6.13 Noise Codes and Ordinances 42	
SECT	ON 7. TOLERANCES ON AIRBORNE SOUND PRESSURE	
	LEVEL ESTIMATES 43	
SECTI	ON 8. VIBRATION ISOLATION OF RECIPROCATING ENGINES 45	
	8.1 On-Grade Location 45	
	8.2 Upper-Floor Location	
	8.3 Special Situations 50	
SECH	ON 9. VIBRATION ISOLATION OF TURBINE ENGINES	
	9.1 On-Grade Location	
	9.2 Upper-Floor Location	
SECTI	ON 10 SUMMARY 53	
SECH	ON 10, SUMMART	
DATA	SHEETS 1-9	
TABL	ES 1-39	

Noise Control for Reciprocating and Turbine Engines driven by Natural Gas and Liquid Fuel

NAL VER AN AND AND THE ALL PARTY OF THE ARTICLE STATE

Ξn .



# **OUTLINE OF MANUAL**

The introduction of the on-site total energy plant represents an important step forward in the continuing program to develop, improve and simplify the uses of natural gas energy in buildings. Other literature on the subject treats the development and uses, economics and growth, and engineering and operation of total energy systems. This manual is devoted to the acoustics—the noise and vibration—of these installations.

In a few of the first installations, enough concern was not given to their noise characteristics, and those installations have been referred to often since then by opponents of total energy. In this manual, recommendations are given for the control of the noise and vibration of these plants by proper architectural, engineering and acoustical design.

The designs cover three typical equipment locations: (1) on a basement or on-grade slab within a building, where there are no acoustically critical occupied areas beneath the engine; (2) at an upper floor location in a building, where there may be acoustically critical occupied areas above, below or beside the engine installation; and (3) in any location where concern may have to be given to the presence of neighbors near the building that houses the equipment.

The driving mechanism for these installations

1

may be either a reciprocating or turbine engine, and it may be fueled by either natural gas or liquid fuel. The driven mechanisms for these installations may be electric generators, refrigeration compressors (either reciprocating or centrifugal), liquid- or gas-transmitting pumps or compressors, or gears that in turn drive some of these devices. In most cases, noise control recommendations given here for the engine will take care of the noise and vibration of the driven mechanism as well.

Because of the fairly general, semi-technical coverage of this manual, many details and qualifications are omitted, and the resulting procedures may not completely solve some of the more complex noise problems associated with these power plants. However, the guidelines offered by the manual will aid in solving most of the typical noise problems that may arise in an ordinary installation.

There are no simple standard solutions to all engine noise problems; that is the reason for this manual. Although at first reading this manual may seem unduly complicated, actual use of the manual is abstracted into the filling-in of several blanks on several special data sheets that lead step-by-step to very specific design decisions. In this way, small engines and large engines, one engine or many engines, a small engine room or a large engine room, a relatively quiet or a relatively noisy

an and the first in a sector of the first of the former all the the sector of the sector of the sector of the s

adjoining area, a nearby or a far-off neighbor, a conservative or a compromise design, a big problem or a little problem, can be solved with almost equal simplicity and versatility. Simple examples along the way help illustrate the step-by-step procedures.

Because the manual is intended to solve problems rather than generate problems, the basic design procedure is aimed at providing a "conservative design" (i.e., one with reasonable assurance of achieving a satisfactory solution) but comments are offered on making some compromise decisions where they may be desired.

In the step-by-step development of the manual procedures, Section 1 outlines the basic noise problem due to engines, describing simply such necessary terms as "sound pressure level" and "sound power level" and presenting a schedule of "noise criteria" used to describe the acoustic environment in most typical living and working areas. Section 2 pertains to the vibration produced by engines and distinguishes between "feelable vibration" and "audible vibration" or "structureborne noise." Section 3 provides quantitative data on the three principal noise sources of reciprocating engines: the casing, the exhaust and the air intake. By means of special data sheets, an example of each source is illus-



Engine, gear and compressor are installed on massive concrete inertia block, which is then supported by steel springs off the floor. This type of installation is required where large reciprocating engines are located on upper floors above acoustically critical areas.

trated. Section 4 provides data and examples of the three principal noise sources of turbine engines. Also, the casing, the exhaust and the air intake.

Section 5 provides a brief introduction into some basic acoustics for indoor situations, including how to estimate sound pressure levels in the engine room, then how to determine how much sound gets into a room adjoining the engine room, either on the same floor or on the floor above or below. This involves the terms "transmission loss" and "noise reduction" of a wall or a floor-ceiling combination.

Several tables of "transmission loss" data are included because the walls and floors become basic design concerns of the architect and mechanical engineer. Certain restrictions on large engines in upper-floor installations are given at the end of Section 5 under the heading "Special Situations."

Section 6 pertains to sound transmission outof-doors and includes the noise reduction effects of distance, barriers and dense woods. Also considered are the directivity of a large exhaust opening and the noise reduction of mufflers for both reciprocating- and turbinetype engines. Section 6 also includes noise reduction involved in indoor noise escaping outdoors and outdoor noise coming indoors.

Section 7 discusses tolerances in airborne noise analysis procedure and offers a schedule for identifying a design as "preferred," "acceptable," "marginal" or "unacceptable." Sections 8 and 9 offer vibration isolation recommendations for reciprocating and turbine engines, respectively. Section 10 serves to summarize noise sources and paths, and emphasizes the need to treat each one in order to achieve a successful installation.

At the end of the text, 39 tables of data provide quantitative material on many aspects of noise control believed to be required for engine noise analysis and solution. Finally, nine data sheets provide step-by-step procedures for carrying out most of the necessary calculations in noise analysis. Copies of these blank data sheets may be used to work out design details of each job.



2 12 10

to any in the weather and the short is not to

It is true that reciprocating and turbine engines may be noisy if nothing is done to control their noise. A bare engine running in a non-acoustically treated engine room could be noisy enough to require that ear protectors be worn by operators in the same room; the noise could be disturbing to nearby occupants in the same building, if the engine room does not adequately confine the noise; and escaping engine noise or engine exhaust noise could even be a source of annoyance to nearby residential neighbors during the quiet of the night.

The degree of disturbance or annoyance of an intruding, unwanted noise depends essentially on three things: (1) the amount and nature of the intruding noise; (2) the amount of back-ground noise already present before the intruding noise occurred; and (3) the nature of the working or living activity of the people occupying the area in which the noise is heard. People trying to sleep in quiet suburban homes would not tolerate very much intruding engine noise, while office workers in a busy mid-city office could have greater amounts of noise without even noticing it; and, finally, factory workers in a continuously noise manufacturing space might not even hear a noisy nearby engine installation.

#### 1.1 Noise Criteria

It is common practice in acoustical engineering to rate various environments by "noise criteria" and to describe these criteria by fairly specific noise level values. Detailed discus-



Hundreds of standby generator sets are located inside telephone company buildings. An adequate generator set enclosure, vibration isolation, and suitable mufflers permit installation in virtually any part of building. (Photograph is courtesy of Solar Division, International Harvester Company.)

sions of noise criteria can be found in other literature<sup>1</sup>, and only a useful summary of that material is introduced here. In the interest of brevity, many important details and qualifications are omitted. Thus, in a complex problem, additional reading or acoustical assistance may be necessary.

4

The second second second

「「「「「「」」」」「「「」」」」」」」

ï

In Table 1, a number of typical indoor living and working spaces are grouped together into "categories." A low category number indicates areas in which relatively low noise levels are desired; higher category numbers indicate areas in which relatively higher noise levels are permissible.

The various indoor areas listed in Table 1 are intended to be illustrative; areas that are not specifically named here can be assigned, from an acoustic point-of-view, to the various categories on the basis of their similarity with the spaces that are named. From time-to-time in the manual, these areas may be referred to as the "noise-sensitive area," the "acoustically critical area" or some other similar term.

Table 1 also includes a "Noise Criterion Designation" for each of these areas. For example, Category 1 areas include "bedrooms, sleeping quarters, hospitals, residences, apartments, hotels, motels, etc. (for sleeping, resting, relaxing)." This group of activity areas is then given the noise criterion designation of NC-20 to NC-30.

#### 1.2 Sound Pressure Level ("SPL")

The ear is sensitive to tiny oscillations of pressure in the air. Sound waves produce these pressure oscillations. The intensity of sound waves may be described quantitatively by the term "sound pressure level" (abbreviated "SPL" in the manual and popularly called sometimes "noise level"). Sound pressure levels used throughout the manual are given in decibels ("dB") relative to the standard reference pressure of 0.0002 microbar, although the reference pressure is often omitted.

Human hearing covers a full frequency range of from about 20 Hz (cps) to 20,000 Hz (cps)<sup>2</sup>. In acoustical engineering applications it is normal practice to divide this frequency range into eight "octaves" (as in "octaves" of a piano scale) and to work with the "octave frequency bands" of noise. The octave frequency bands that are in common usage now are identified by their "center frequencies" as follows: 63 Hz, 125, 250, 500, 1,000, 2,000, 4,000, and 8,000 Hz. Actually, each of these frequency," and the actual bandwidth extends from 0.7 of the center frequency to 1.4 times the center frequency. An octave band centered at 31 cps is sometimes used, but it is not included here.

In Table 2, actual sound pressure levels (SPL) are listed for the eight octave frequency bands for each of the noise criterion (NC) designations used in Table 1. In later applications within the manual, it will be the goal of a particular "noise control design" to meet or not exceed the noise levels of the appropriate NC designation for a given critical area. In applying the indoor NC designations of Table 1, the lower limit of the range for a given category (for example, NC-20 for Category 1) should be used for a critical situation, while the upper limit (or some intermediate value) may be used for a situation known to be not so critical.

A special note of concern is given for the Category 1 and 2 areas of Table 1. For a very quiet community area or for a quiet building with no internal ventilation system noise, the NC-20 noise criterion should be applied for indoor conditions. For a noisy city environment outdoors or for a building with a ventilation system known to fall in the NC-30 noise range, an NC-30 noise criterion can be applied

l yn enner Diensende Bereiken ferener en en en selferer i fer en fer en skere en skere en se en se i der i der o

<sup>&</sup>lt;sup>1</sup> For a quantitative discussion of noise criteria and noise levels, refer to a textbook or reference book on acoustics, such as "Noise Reduction," Leo L. Beranek, Editor, McGraw-Hill Book Company (1960), "Handbook of Noise Control," C.M. Harris, Editor, McGraw-Hill Book Company (1954), the latest issue of the ASHRAE "Guide and Data Book," American Society of Heating, Refrigeration and Air-Conditioning Engineers, Inc., 345 East 47th Street, New York, N.Y. 10017 or selected topics of the Journal of the Acoustical Society of America.

<sup>&</sup>lt;sup>2</sup> The recently accepted U.S. and international standard unit of frequency is the "Hertz," abbreviated "Hz." Thus, Hertz has the same meaning and value as the traditional and familiar term "cycles per second" or "cps." The new unit Hz is used in this manual.

to rooms other than bedrooms or auditoriums. For bedrooms or auditoriums or for situations that do not clearly fall at the NC-20 lower limit or NC-30 upper limit, NC-25 indoor noise criterion levels should be applied.

#### 1.3 Sound Power Level ("PWL")

The sound pressure level in a room is a function of (1) the "acoustics" of the room and (2) the strength of the sound source. The strength of the source may be described quantitatively as the "sound power" radiated by the source, and, when used in the proper units, this becomes "sound power level" (abbreviated "PWL" in this manual), expressed in dB relative to the reference power of  $10^{-12}$ watt<sup>3</sup>. In Sections 3 and 4 of the manual, sound power level data are given for reciprocating and turbine engines, and engine components. As a simple analogy, "sound power" of a source might be likened to the power rating for an electric lamp (e.g., "100-watt" bulb), while "sound pressure" might be likened to the illumination in a room provided by that electric lamp. This analogy also helps explain the role of acoustic absorption in a room.

In a room with white-painted walls and ceiling, a weak light source will give fairly good general illumination, but if the room had only dark, unreflecting surfaces, the general illumination would be very poor. Similarly, a sound *power* source may produce high or low sound *pressure* levels in a room depending on the degree of absorption or reflection of the surfaces inside the room. The acoustics of the room are discussed more quantitatively in Section 5.

<sup>&</sup>lt;sup>3</sup>For a more technical description of PWL and SPL refer to an acoustics textbook.

# Section 2 THE VIBRATION PROBLEM

The vibration produced in a building by an operating machine can manifest itself in two ways to a person in that building: (1) by being "feelable" and (2) by being "audible." Building structures and the earth are good transmitters of vibrational energy, and if the transmitted vibration is intense enough, it will be "feelable" as vibration. If the vibration is not intense enough to be "feelable," it may still be "audible."

In a very quiet environment, a large vibrating surface can radiate "audible" sound when its vibration level is only 1/100th to 1/1,000th of that required to be feelable. In fact, a wall that vibrates with an *unfeelable* amplitude of approximately one millionth of an inch can still be *heard* to radiate sound in a very quiet room.

Thus, it is not enough just to prevent an engine from producing feelable vibration in a building; it is usually necessary to reduce building vibration to a level much lower than this in order that building occupants will not hear the noise radiated by the vibrating walls, floors, ceilings, columns, beams, doors, windows, etc. It is this type of vibration that is known as "structure-borne noise."

When designing or specifying a vibration isolation mounting assembly for an engine, the

- THE REPORT OF A CONTRACT OF A C

acoustic objective of that design is to reduce the "structure-borne noise" so that it will be considered acceptable in those various categories of spaces given in Table 1. This becomes a particularly difficult requirement when a reciprocating engine is mounted on the upper floor of a building, because the low-speed, unbalanced forces of the reciprocating engine may "drive" the building elements (walls, beams, floors, etc.) at or near their natural resonance frequencies and produce audible or feelable vibration at remote parts of the building away from the engine.

In some special-purpose buildings there may be some vibration-sensitive equipment, instru-

ments or processes that are more sensitive than people to the background vibration in the building. Some medical, magnetic, electric and electronic equipment may be subject to this low-level vibration pickup. It is usually necessary to provide special isolation mountings on this type of equipment so that it will not be sensitive to normal building vibration, whether caused by mechanical equipment or by occupants of the building (walking, closing doors, typing, etc.). In this manual, engine mountings will aim at achieving acceptable vibration and structure-borne noise levels in the building to meet the requirements of human occupants. Special mountings to meet the needs of vibration-sensitive equipment are beyond the scope of the manual.



In this arrangement of generator sets, each assembly is supported on pad mounts at the floor and includes flexible connections to overhead air intake duct, exhaust mufflers and power outlet. (Photograph is courtesy of Caterpillar Tractor Company.)

Section 3 RECIPROCATING ENGINE NOISE

「「いたい」となった。 いいで かいしん

For the preparation of this manual, noise data were obtained and studied for over 75 reciprocating natural gas and diesel-fuel engines covering a power range of 10 to 6,000 hp. For the purposes of the manual, interest is limited to reciprocating engines in the power range of 20 to 2,500 hp.

Measured noise data have been studied extensively to determine approximately the influence of certain operational variables on the amount of noise generated by these engines. From this study it now appears possible to predict fairly accurately the amount of noise generated by any reciprocating engine fueled by natural gas or liquid fuel. In this manual, technical details of the study and the results are not given, but useful conclusions are summarized and used<sup>1</sup>. For reciprocating engines, there are three principal noise sources of concern: (1) the engine casing; (2) the engine exhaust; and (3) the air intake to the engine. These noise sources are now considered one at a time.

9

en ig er som ande ande i Dielen af en en som en en som en som som som som en som en som en som en som en som en

<sup>&</sup>lt;sup>1</sup>Two concurrent but independent projects, carried out by Bolt Beranek and Newman Inc. during 1966-1967, provided the noise data used in this study. The first of these projects was sponsored by the Office of the Chief of Engineers of the Department of the army. One part of that project was the preparation of the engineering manual "Power Plant Acoustics: Noise Control for Diesel, Gas and Gas Turbine Engine Installations" by Laymon N. Miller, September 1967. That manual summarizes noise data in technical detail. The second project has been sponsored by the American Gas Association and has led to the preparation of this manual. This manual minimizes the technical detail of the noise data but attempts to use the data for practical engineering and architectural purposes.

D-4-0							
Data Si	ieet 1				,	(Paragr	aph 3.2 Exa
			Estimated S Reciproc	Sound Power ating Engine	Level (PWL) of Casing Noise	ſ	
1. Conti	uous Ratin	g of Engine;	200	hp or	k	w	
2. Engin	speed:	1,200	rpm				
3. Fuel:	Gas Only	Z Liqu	id Onlý 🛛	Gas and Liq	uld 🗖		
[			<u> </u>	Frequency Bar	nd in Hz		
63	125	250	500	1,000	2,000	4,000	8,00
4. Base F	WL from Ta	ble 3 for Ite	IO8	108	107	101	94
5. Speed	Correction f	rom Table 3	for Item 2 S	peed:			
	-2	-2	-2	-2	-2	-2	-2
-2	prrection fre	nn Table 3 f	or Item 3 Fue	11:			
6. Fuel C			-3	-3	-3	-3	-3
6. Fuel C	-3	<u> </u>					
6. Fuel C -3 7. Estima Cautio	-3 ted PWL (in 1: Observe a	dB re 10-1 Igebraic sign	<sup>2</sup> watt) of Ca: is in combinin	sing Noise: (Iter g itemsl	n 7 = (tem 4 + lt	em 5 + Item 6)	

#### 3.1 Engine Casing Noise

In terms of a rather simplified approach, the noise radiated by the engine casing is related to the following influencing factors: (1) continuous rating of the engine in hp (horse $power)^2$ , (2) shaft speed of engine, and (3) type of fuel used by the engine. Table 3, at the end of the text, summarizes the procedure for estimating the sound power level (PWL) of an engine casing for a particular rating, speed and fuel combination. This does not include the noise of the exhaust from the engine nor the noise of the exhaust from the engine or of the larger engines that have turbochargers. As seen in Table 3, the "Estimated PWL" is made up of a "Base PWL" plus two correction terms. The Base PWL values differ for the eight octave frequency bands, but the correction terms apply equally to all bands.

## 3.2 Example, Data Sheet 1

As an example of the use of Table 3, suppose that it is desired to know the approximate PWL of a 200-hp, 1,200-rpm natural gas engine used to drive an electric generator. Sample Data Sheet 1 indicates the steps taken to make the calculation. For this example, each blank of Data Sheet 1 is filled in with the appropriate material from Table 3. (A blank copy of each data sheet used in the manual is given at the end of the text. Reproduced copies of these data sheets may be used for specific design problems.)

#### 3.3 Engine Exhaust Noise

The noise generated by the unmuffled engine exhaust and radiated from the open end of the exhaust pipe is related to the following influencing factors: (1) continuous rating of the engine in hp (horsepower) or kw (kilowatts); (2) the presence or absence of an air intake turbocharger driven by the exhaust gases; and (3) the length of the exhaust pipe leading exhaust gases away from the engine. Table 4 summarizes the data used for estimat-

<sup>2</sup>To convert hp (horsepower) output to kw (kilowatts) produced, or vice versa, use the relationship: kw = hpl l.5

and the second second

ing the PWL of the exhaust noise. When a turbocharger is driven by the released exhaust gases, some energy is removed from the exhaust, and less noise is produced. A nominal reduction of 6 dB is shown in Table 4 for this effect.

A long exhaust pipe also helps reduce the escaping exhaust noise, approximately as shown by the correction term in Table 4. It is not suggested here, however, that a long exhaust pipe may take the place of a muffler as a noise control device.

It is cautioned that Table 4 data may not accurately predict the low frequency exhaust noise of some of the large low-speed engines. In the cases of some of these engines, the muffler manufacturers design special mufflers to control the exhaust noise.

#### 3.4 Example, Data Sheet 2

Suppose that the 200-hp engine of Example 3.2 is fitted with a turbocharger and a



Ready Power chiller assembly includes flexible connections in piping and vibration isolation mounts at floor. When enclosed springs are used, as here, check carefully that mounting system is properly aligned and that springs are not binding or shorted out. (Photograph is courtesy of The Ready Power Company.)

Data Sheet 2

(Paragraph 3,4 Example)

# Estimated Sound Power Level (PWL) of Unmuffled Reciprocating Engine Exhaust Noise

1. Continuous Rating of Engine:\_\_\_\_\_\_hp\_or\_\_\_\_\_kw

2. Air Intake: Turbocharger 🖾, 👘 No Turbocharger 🗖

3. Exhaust Pipe Length: 20 ft.

	Frequency Band in Hz										
63	125	250	500	1,000	2,000	4,000	8,000				

## 4. Base PWL from Table 4 for Item 1 Rating:

				·	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	
132	138	134	126	122	116	106	98
	<u>I</u>	L	[	·	·	L	L.,,

5. Turbocharger Correction from Table 4 for Item 2 Air Intake:

-6	-6	-6	6	-6	-6	-6	-6
L		[		<u> </u>		L	L

6. Exhaust Pipe Length Correction from Table 4 for Item 3 Length:

	-5	-5	-5	5	-5	-5	-5	-5
ļ			L		l	l	<u>ا</u>	l

 Estimated PWL (in dB re 10<sup>-12</sup> watt) of Unmuffled Exhaust Noise: (Item 7 = Item 4 + Item 5 + Item 6) Caution: Observe algebraic signs in combining items!

121	127	123	115	111	105	95	87
		└────	,,	,			

(Paragraph 3.6 Example)

### Estimated Sound Power Level (PWL) of Untreated Turbocharger Noise at Air Inlet Opening of Reciprocating Engine

1. Continuous Rating of Engine: 2,000 hp or kw

2, Inlet Air Duct Length: \_\_\_\_\_30 \_\_\_\_ft.

Data Sheet 3

	Frequency Band in Hz									
63	125	250	500	1,000	2,000	4,000	8,000			

3. Base PWL from Table 5 for item 1 Rating:

	100	98	98	99	102	103	102	94
ļ		L				L	I	

4. Inlet Air Duct Length Correction from Table 5 for Item 2 Length:

1								
		-	_	_	_	_	_	
	~b	- 5	-5		-5	-5	-5	-5
		-	-				-/	

5. Estimated PWL (in dB re  $10^{-12}$  watt) of Untreated Turbocharger Noise: (item 5 = item 3 + item 4) Caution: Observe algebraic signs in combining items!

95	93	93	94	97	. 98	97	89
the second se					·	[	

States, See

The second second of the second se

APRO FARATON

20-foot-long exhaust pipe leading to the roof of the engine room. Sample Data Sheet 2 summarizes the steps involved in calculating the PWL of the unmufiled exhaust using data from Table 4.

#### 3.5 Engine Air-Intake Noise

Some of the smaller engines covered by this manual do not have turbochargers; instead, the intake air is drawn into the engine by "natural aspiration" (designated by "NA" in some engine catalogs). On the other hand, many of the smaller- and medium-sized engines are provided with turbochargers (a turbine-type device, driven at high speed by the released exhaust gases, that increases the flow and pressure of air into the engine). These smaller turbochargers are usually installed directly in the short air-intake duct to the engine just beyond an air-intake cleaner or filter.

The high frequency white of the turbocharger is usually masked by the total noise of the engine, although it can be heard up close to the air cleaner opening. The noise of these turbochargers may be assumed to be contained within the total noise of the engine, and no special noise control treatment will be required—other than a caution against listening with the unprotected ear directly at the air cleaner inlet.

For many of the larger and slower engines (say, over 1,000 hp, although there is no clear size distinction), the manufacturer may specify a ducted connection to bring room air or outside air into the engine intake and turbocharger. In this event, the turbocharger noise radiated into the room or out-of-doors may be of concern. A rough estimate of the sound power level of the escaping turbocharger noise is given in Table 5. It is cautioned here also that this may not hold true for all engines and turbochargers. It is important, however, to recognize that turbocharger noise should be considered and that Table 5 data will provide a reasonable approximation of this noise.

#### 3.6 Example, Data Sheet 3

As a sample calculation using Table 5, suppose that a turbocharged 2,000-hp engine has a 30-foot ducted air inlet from outside the engine room. Sample Data Sheet 3 is used to carry out the steps of this calculation. These values are not claimed to be highly accurate because different turbochargers make different amounts of noise, depending on design variables. And the inlet duct may reduce the escaping noise by different amounts than that given in Table 5. Nevertheless, the estimate may serve as a guideline for potential noise.

#### 3.7 Vibration Data

Vibration data have been taken on several of the reciprocating engines studied in this program; however, because of the many variables in engines, mounting arrangements and mounting locations and because vibration data are not readily understood or appreciated by most non-acousticians, summaries of the data are not given here. In Section 8 of the manual, vibration isolation recommendations for various types of reciprocating engine installations are given. The most critical installations, of course, are those located on upper floors of buildings and directly over quiet occupied spaces.

Section 4

For the preparation of this manual, noise data were obtained and studied for over 45 turbine engines covering a power range of 200 to 20,000 kw. For the purposes of the manual, interest is limited to the power range of 200 to 5,000 kw. 300

From the study of turbine engine noise data, it is possible to estimate for engineering purposes the noise from three principal sources: (1) engine casing, (2) engine exhaust, and (3) air intake to the engine. These three noise sources are now considered one at a time.

4.1 Engine Casing Noise

The approximate PWL values for the casing noise of unenclosed, untreated gas turbine engines are given in Table  $6^1$ . These values apply to a bare engine having no thermal insulating cover or form of enclosing cabinet.

If there is some form of cover for the entire engine casing, the radiated noise will probably be reduced, depending on the acoustic quality

of the cover. The footnote of Table 6 lists a few typical covers that may be provided with some engines and suggests an approximate noise correction that might be attributed to each general type of cover.  $-7.500 h \rho$ 

#### 4.2 Engine Exhaust Noise

The approximate PWL values for the unmuffled exhaust noise from gas turbine en-



AiResearch turbine engine-generator set is packaged inside cabinet for shipment and later installation at computer center. Exhaust heat from engines is used to operate absorption refrigeration machine which air conditions computer installation. (Photograph is courtesy of Garrett-AiResearch Manufacturing Division.)

1.117.11

1. 1.

<sup>&</sup>lt;sup>1</sup>Turbine engine ratings are usually quoted in kilowatt or megawatt values. To convert kilowatts (kw) output and horsepower (hp) produced, use the relationship: kw = hp/1.5

gines are given in Table 7. These values apply to the noise radiated directly from the end of the engine, assuming no exhaust duct. Actually, most turbine engine installations will have connecting ducts from the engine exhaust to some outdoor opening, and these ducts may be counted on for some degree of silencing, depending on the nature of the acoustic or thermal lining of the duct.

THE REAL PROPERTY OF A DESCRIPTION OF A

The approximate attenuation provided by straight ducts and duct turns for a typical engine exhaust duct may be estimated from data in Tables 9 and 10 or from data in the "ASHRAE Guide"<sup>2</sup>. Attention is directed to the qualifying comments in Tables 9 and 10. It is especially important to realize that, at elevated exhaust temperatures, the wavelength of sound in the exhaust gas is longer than at ordinary temperature, and, as a result, the duct appears shorter, in term's of wavelengths, to the sound wave. Thus, the attenuation will be less. As a first approximation, it is suggested that two-thirds of the normal attenuation be used when hot exhaust ducts are being considered. It is hardly necessary to be more exacting here for estimates of duct losses. It is more important later to take this temperature effect into account when considering mufflers for controlling the noise.

If the turbine engine manufacturer supplies with the engine an exhaust muffler of known insertion loss, the PWL values of Table 7 may be reduced by the muffler insertion loss values for each of the frequency bands (corrected for exhaust temperature). Provision for this is made in the appropriate data sheet.

#### 4.3 Engine Air Intake Noise

The approximate PWL values for the unsilenced air-intake noise of gas turbine engines are given in Table 8. These values apply to the noise radiated directly from the air intake end of the engine. If an inlet duct brings air into the engine, the approximate sound attenuation provided by the duct can be estimated from Tables 9 and 10 or from the "ASHRAE Guide." Please note the qualifications associated with data of Tables 9 and 10.

If the engine manufacturer supplies an airintake silencer of known insertion loss, the PWL values of Table 8 may be reduced by the silencer insertion loss values for each of the frequency bands,

#### 4.4 Example, Data Sheet 4

**法运行**能力的利益

Sta & C & E &

Suppose a 2,500-kw turbine-driven generator set is to be installed in a building. As supplied by the manufacturer, the packaged set is housed in a metal cabinet with inside acoustic lining but with many open holes for ventilation. The circular exhaust duct is 30 feet long and extends to the roof of the building. This duct is wrapped on the outside with thermal insulation. The exhaust duct also contains a packaged exhaust muffler which has the following insertion loss, corrected for exhaust temperature, according to the muffler manfacturer:

Hz	63	125	250	500	1,000	2,000	4,000	8,000
PWL, dB	5	10	13	15	19	15	12	10

The air intake to the engine is provided by a 12-foot-long rectangular duct to the outside wall of the building. A 2-inch-thick internal acoustic lining is offered by the manufacturer for this intake duct. It is desired to know the PWL for the casing, exhaust and intake for this unit.

Sample Data Sheet 4 summarizes the steps for arriving at the desired data, making use of the material from Tables 6-10. The desired PWL values for the three principal noise sources are given in Items 9, 13 and 17 of Data Sheet 4 after the appropriate corrections are made for the various noise reduction treatments supplied for this particular installation.

#### 4.5 Vibration Data

Because of the rotary action of the turbine engine, vibration is much lower than for reciprocating engines. However, structure-borne noise may still be of concern in some installations, and suggestions for vibration isolation are given in Section 9 of the manual.

<sup>&</sup>lt;sup>2</sup>Refer to the latest available "ASHRAE Guide and Data Book," published by the American Society of Heating, Refrigeration and Air-Conditioning Engineers, 345 East 47th Street, New York, New York 10017.

		Ea	timeted Co.	und Damas Tr	nut (BMIL) - 6	Carling	
		Ex	haust and Ir	itake Noise o	f Gas Turbine	Engine	
1. Cor	ntinuous Ra	ting of Engine	:		2,500	_kw	
2. Eng	ine Casing	Cover: None	🗋 ог Туре	<u>l</u> 4	from Table 6		
3. Ext	aust Duct:	Round 🖾,	Rectangular	🗇, Length	30	_ft,	
Dut	t Lining:	Туре2	<u>?</u> f	rom Table 9			
Duc	t Turns;	No	1	Гуре	from 'Ta	ble 10	
4. Inta	ke Duct:	Round 🗆,	Rectangular	🖾, Length_	12	_ft,	
Duc	t Lining:	Type1	∔f	rom Table 9		1	
Dud	t Turns:	No	рт	`уре	from Ta	ble 10	
6. Inta	ke Muffler (	Supplied: Y	′es 🗋 🛛 N	io 🖾			
6. Inta	ke Muffler :	Supplied: Y	/es [] N	Frequency Ba	nd in Hz		
6. Inta	ke Muffler 1 3 125	Supplied: Y	Yes [] N	Frequency Ba	ind in Hz 2,000	4,000	8,000
6. Inta 6. Inta 7. PWL 11.	ke Muffler ( 3 125 (in dB re 1) 5 118	Supplied: Y 5 250 D <sup>-12</sup> watt) of 119	res [] N 500 Casing Noise 119	frequency Ba 1,000 from Table 6 fo 119	or Itern 1 Rating	4,000	8,000
6. Inta 6. Inta 7. PWL 11. 8. Nois	ke Muffler 1 3 125 (in dB re 1 5 218 9 Reduction	Supplied: Y	res [] N 500 Casing Nolse 119 Casing Cover (	Frequency Ba 1,000 from Table 6 fo 119	nd in Hz 2,000 or Itern 1 Rating 119 n 2 as Estimated	4,000 : 119 in Table 6 Footn	8,000 119 ote:
6. Inta 6. Inta 7. PWL 11. 8. Nois	ke Muffler : 3 125 (in dB re 1) 5 218 9 Reduction 4 -14	Supplied: Y	res I N 500 Casing Noise 119 Casing Cover 1 -6	Frequency Ba Frequency Ba 1,000 from Table 6 fo 119 (If Any) of Iter -7	ind in Hz 2,000 or Item 1 Rating 119 n 2 as Estimated ~8	4,000 : 119 in Table 6 Footn 8	8,000 119 ote: -8
6. Inta 6. Inta 6 7. PWL 11. 8. Nois: 8. Nois: 9. PWL Caut	ke Muffler 3 122 (in dB re 1) 5 118 9 Reduction 9 -4 of Casing N an: Observe	Supplied: Y	res D N 500 Casing Noise 119 Casing Cover ( -6 er (If Any) Ito is in combining	Frequency Ba Frequency Ba 1,000 from Table 6 fo 119 (If Any) of Iter -7 am 9 = Item 7 m ig items1	ind in Hz 2,000 or Itern 1 Rating 119 n 2 as Estimated ~8 + Item 8	4,000 : 119 in Table 6 Footn 8	8,000 119 ote: -8
6. Inta 6. Inta 6 7. PWL 111 8. Noise 8. Noise 9. PWL Cauti 112	ke Muffler           3         128           (in dB re 1)           5         118           e Reduction           -14           of Casing N           on: Observe           114	Supplied: Y	Yes     N       500       Casing Noise       119       Casing Cover       -6       er (If Any) Iters in combining       113	Io II           Frequency Ba           1,000           from Table 6 fr           119           (If Any) of Iter           -7           am 9 = Item 7 mg items!           112	ind in Hz 2,000 or I tern 1 Rating 119 n 2 as Estimated -8 H Item 8 111	4,000 : 119 in Table 6 Footn 8	8,000 119 ote: -8

(Paragraph 4.4 Example)

Sheet 2 of 2

### Data Sheet 4 (continued)

				Frequency Ban	d in Hz		
63	125	250	500	1,000	2,000	4,000	8,000

10, PWL (in dB re 10<sup>-12</sup> watt) of Exhaust Noise from Table 7 for Item 1 Rating:

		r	·		·	·	· · · · · · · · · · · · · · · · · · ·	<del></del>
	130	132	132	131	129	127	123	117
- 1								

11. Attenuation Provided by Exhaust Duct and Turns of Item 3, from Tables 9 and 10 or "ASHRAE Guide" (include hot temperature correction):

İ	-3	-2	-1	-0	-0	-0	-0	0
		1			L		L	

12. Insertion Loss of Exhaust Muffler or Item 5, If Provided (use muffler manufacturer's data for appropriate bands, corrected for exhaust temperature):

Ì	5	-10	-13	-15	-19	-15	-12	-10	
ļ	نىمىسىك			Ĺ	<del>_</del>		L	L	_

13. PWL of Exhaust Noise Out of Duct and Muffler (as applicable): Item 13 = Item 10 + Item 11 + Item 12 Caution: Observe algebraic signs in combining Items!

					·		
122	120	118	116	110	112	111	107
			-+-				
			ليرجي ويستعجبها				

14, PWL (in dB re 10<sup>-12</sup> watt) of Air-Intake Noise from Table B for Item 1 Rating:

			······································	r		I	
117	118	118	121	127	132	132	129
			<u> </u>	I	I	L	L

15. Attenuation Provided by Intake Duct and Turns of Item 4, from Tables 9 and 10 or "ASHRAE Guide":

	-4	-3	-5	-11	-13	-13	-11	-7
1		<u> </u>		L	L	L	·	<u></u>

 Insertion Loss of Intake Muffler of Item 6, If Provided (use muffler manufacturer's data for appropriate bands; see Table 39):

1	1 7			1

17. PWL of Intake Noise Out of Duct and Muffler (as Applicable) Item 17 = Item 14 + Item 15 + Item 16 Caution: Observe algebraic signs in combining items!

113	115	113	110	114	119	121	122

Section 5 CONTROL OF AIRBORNE SOUND INDOORS

Principal concern with the noise of enginegenerator sets is the airborne sound radiated directly from the noise-making components. As the term "airborne" suggests, this sound is carried essentially by air paths from the source to the receiver. In some cases the noise may strike a massive wall and even be transmitted "through" the wall to the opposite side where it continues to be radiated and considered as airborne noise.

The control of airborne sound involves: (1) some fundamentals about the distribution of sound inside rooms and outdoors, (2) some quantitative data on the ability of a wall to reduce sound transmission, and (3) some quantitative data on the effectiveness of sound-attenuating mufflers placed in sound passages. Indoor aspects of airborne sound are discussed in this section, while outdoor aspects are considered in Section 6.

#### 5.1 Sound Pressure Level in a Room

「「たいとないない」」というという

In Paragraph 1.3 it was mentioned that the sound pressure level (SPL or noise level) in a room is a function of (1) the acoustics of the room, and (2) the strength of the source. In Sections 3 and 4, data have been given on the "strength of the source," namely, the sound power level (PWL) of the major noise sources of reciprocating and turbine engines. Now, in this part of Section 5, the acoustics of a room are considered.

In a brief and simplified procedure, the acous-

The second 
tics of a room may be described by the volume of the room and the relative amount of "acoustic absorption" or "acoustic treatment" in the room. Then, the SPL (sound pressure level) in the room can be estimated for any PWL (sound power level) in the room by applying a "room correction term" that takes into account the volume and acoustic absorption of the room. Thus,

#### SPL = PWL + Room Correction Term

Table 11 gives average values of the "room correction term" that will yield the "reverberant" noise level in a room, that is, the noise level that may be found to exist fairly uniformly all around the room at a distance of more than about 5 to 10 feet from the source for smaller rooms and more than about 10 to 20 feet from the source for larger rooms. At closer distances to the noise source, the noise will be higher than this "reverberant" noise level by a few decibels, depending on a number of details which will not be discussed here. As a rough approximation of the SPL within a few feet of an engine, the reverberant SPL can be increased by 3 dB. The room correction term actually varies with frequency but that variation is ignored here and the values given in Table 11 may be applied equally to all octave frequency bands. For a more rigorous and detailed analysis of a room acoustics problem, the reader is referred to a textbook on acoustics.

The footnotes of Table 11 describe four dif-

ferent degrees of room treatments that might be used. Various areas and thicknesses of acoustic absorption material are suggested. Where a 3/4 inch to 1 inch thick absorption material is used, it is intended that such a material have a published "NRC" value ("noise reduction coefficient") in the range of 0.65 to 0.75. Where a 1-1/2 inch to 2 inch thick absorption material is used it is intended that such a material have a published NRC value in the range of 0.75 to 0.85. Bulletins on acoustic absorption materials list all of the marketed products that meet these values<sup>1</sup>.

#### 5.2 Example, Data Sheet 5

As an example of the use of Table 11, suppose that the 200-hp, 1,200-rpm natural gas engine of Example 3.2 is to drive an electric generator. Suppose the engine room is 12 feet high, 15 feet wide and 20 feet long. On the underside of the roof there is a 2-inch-thick layer of exposed glass fiber insulation that serves both as thermal insulation and as acoustic absorption. It is desired to know the estimated "noise levels" (sound pressure levels) in the room when the engine is running.

To aid in the estimating procedure, Data Sheet 5 may be used. This sheet indicates each of the steps necessary to determine the estimated reverberant SPL in the engine room. For this example, each blank of Data Sheet 5 is filled in with the appropriate material. The Item 8 PWL values are taken from sample Data Sheet 1 used with Example 3.2. Item 9 gives the reverberant SPL in the engine room, and Item 10 gives the approximate SPL near an engine

#### 5.3 Addition of Decibel Levels

When several engines or noise sources are present in a given space, the total sound power level or the total noise level will usually be greater than that for any single engine or

noise source in the space. It is necessary to take this into account in any particular noise control design, especially since there frequently are several engines operating in the same room. Since PWL and SPL values are logarithmic-based numbers, they cannot be added together by simple algebraic addition. For example, 90 dB + 90 dB does not give 180 dB, instead, as in electrical power addition of decibels, 90 dB + 90 dB = 93 dB, i.e., two equal sources added together produce a total level that is 3 dB greater than the level of either source. Table 12 summarizes four simple rules to be followed for adding SPL or PWL contributions to obtain the total SPL or PWL.

As illustrations of the use of Table 12, according to the first rule,

$$95 dB + 97 dB = 99 dB$$
  
 $90 dB + 97 dB = 98 dB$   
 $85 dB + 95 dB = 95 dB$ 

According to the second rule,

80 dB + 80 dB = 83 dB95 dB + 95 dB + 95 dB = 100 dB.

To illustrate the third rule, the following four sample SPL values are added by using two different orders of addition:

First order,

$$\begin{cases} 88 \ dB \\ 90 \ dB \\ 93 \ dB \\ 98 \ dB \\ 99 \ dB \\ \end{cases} = 92 \ dB \\ = 100 \ dB \\ = 100 \ dB$$

Second order,

$$\begin{cases} 88 \ dB \\ 98 \ dB \\ 90 \ dB \\ 93 \ dB \\ \end{cases} = 95 \ dB \\ = 95 \ dB \\ \end{cases} = 100 \ dB$$

## 5.4 Transmission Loss of Walls ("TL")

When a sound wave strikes the "front" surface of a solid wall, there is enough energy in the tiny pressure oscillations in the air to cause the whole wall to vibrate. In vibrating,

<sup>&</sup>lt;sup>1</sup>Refer to manufacturers' published data or see latest applicable annual bulletin of "Performance Data of Architectural Acoustics Materials" or its successor, published by the Acoustical Materials Association, 335 East 45th Street, New York, New York 10017, or its successor.

ita She	et 5					(Paragri	aph 5.2 Example)
		Estimate	d Sound Pr	essure Level (	SPL) in an En	gine Room	
Volum	e of Engine	Room:	2ft, X	<u>15</u> ft.X	20ft. = _3,0	500_cu, ft,	
Total In	nterior Surf	ace Area of	Room (inclu	ding floor) = <u>1</u>	. <u>,440</u> sq, ft.		
, Area of	Acoustic T	reatment_	<u>300</u> sq. f	t.			
. Per cen	t Area Cove	red by Aco	ustic Treatme	ent (item 3/iter	m 2) X 100 =2	2%	
Thicks	act of Abro	ntion Mate	rial 3/4 in - 1	in ["] 1-1/2 in	-2 in X		
Inickne	ess of Absu	рион ман	nai 2/4 in.— i	in, ∟i 1-1/2 in.	2 m, 4:1		
. Acousti Conditi	ic Treatmen on 1 🗖	t Condition Conditio	of Room fro	om Items 4 and 1 Condition 3 🖾	δ, as Defined in i Condition	Footnotes of Table	B 11:
. Acousti Conditi	ic Treatmen on 1 🖸	t Condition Conditio	of Room fro n 2 🔲 🛛	om Items 4 and 1 Condition 3 🖾	5, as Defined in I Condition	Footnotes of Table	B 11:
Acousti Conditi Room C	ic Treatmen on 1 🖸 Correction 1	t Condition Conditio Ferm from 1	of Room fro n 2 🗆 ( Table 11 for l	tems 1 and 6;	5, as Defined in I Condition 8 d in Hz	Footnotes of Table 4 D	s 11:
Acousti Conditi Room C 63	ic Treatmen on 1 🖸 Correction 1	t Condition Conditio Ferm from 1 	of Room fro n 2 🗌 n Table 11 for I 500	tems 1 and 6: Frequency Bar 1,000	5, as Defined in 1 Condition 8 d in Hz 2,000	eoothotes of Table 4 D dB 4,000	a 11: 8,000
Acousti Conditi Room C 63 Total Pi from Ite	ic Treatmen on 1 Correction 7 125 WL of All N em 9 of Dat	t Condition Conditio Ferm from 1 250 Loise Source a Sheet 4 fo	of Room fro n 2 Table 11 for I 500 s in Engine F or Turbine Er	The second secon	5, as Defined in a Condition 8 id in Hz 2,000 n 7 of Data Shee	aothotes of Table 4 [] _dB 4,000 t 1 for Reciprocat	8,000 9,000
Acousti Conditi Room C 63 Total Pt from Ite 100	ic Treatmen on 1 Correction 7 125 WL of All N em 9 of Dat 104	t Condition Conditio Ferm from 1 250 loise Source a Sheet 4 fc 104	of Room fro n 2 Table 11 for I 500 s in Engine F Turbine Er 103	terns 1 and 6; Frequency Bar 1,000 foom (from Iten ngines): 103	5, as Defined in 1 Condition 8 d in Hz 2,000 n 7 of Data Shee 102	A C 4 4 4 4 4 5 4 5 5 5 5 5 5 5 5 5 5 5 5	8,000 ing Engines or 89
Acousti Conditi Room C 63 Total Pt from Ito 1.00 Reverbe and Is Z Caution	Correction 1 Correction 1 125 WL of All N em 9 of Data 104 crant SPL in tero or a Net : Observe a	t Condition Conditio Ferm from 1 250 250 Ioise Source a Sheet 4 fo 104 Engine Ro getive Quan Igebraic sign	of Room fro n 2 Table 11 for I 500 s in Engine F or Turbine Er 103 om (Item 9 = tity for All F is in combini	tems 1 and 6: Frequency Bar 1,000 foom (from Iten ngines): 103 ftem 8 + Item 1 foom Condition ng terms!	5, as Defined in 1 Condition -8 d in Hz 2,000 n 7 of Data Shee 102 7; Item 7 Value I s)	A CO A CO A CO A CO A CO A CO A CO A CO	8,000 ing Engines or 89 equency Bands
Acousti Conditi Room C 63 Total Pt from Ito 100 Reverbe and Is Z Caution 92	Correction 1 125 WL of All N em 9 of Dat 104 Prant SPL in tero or a Net : Observe a 96	t Condition Conditio Ferm from 1 250 loise Source a Sheet 4 fc 104 Engine Ro gative Quar Igebraic sign	of Room fro n 2 Table 11 for 1 500 s in Engine F or Turbine Er 103 om (Item 9 = tity for All F is in combini	terns 1 and 6: Frequency Bar 1,000 1	5, as Defined in 1 Condition -8 d in Hz 2,000 n 7 of Data Shee 102 7; Item 7 Value I s) 94	A CO A CO A CO A CO A CO A CO A CO A CO	8,000 ing Engines or 89 equency Bands
Acousti Conditi Room C 83 Total Pi from Ito 100 Reverbe and Is Z Caution 92 Approxi	ic Treatmen on 1 Correction 7 125 WL of All N em 9 of Dat 104 crant SPL ir Cero or a Ne : Observe a 96	t Condition Conditio Ferm from 1 250 Ioise Source a Sheet 4 fc 104 Engine Ro getive Quan Igebraic sign 96	of Room fro n 2 Table 11 for 1 500 s in Engine F 500 s in Engine F 103 om (Item 9 = tity for All F is in combini 95 (Item 10 = 1	tems 1 and 6: Frequency Bar 1,000 1	5, as Defined in a Condition 8 id in Hz 2,000 n 7 of Data Shee 102 7; Item 7 Value 1 s) 94	A CO A CO A CO A CO A CO A CO A CO A CO	8,000 ing Engines or 89 equency Bands 81

Management and the

.

and the second second

this wall sets into oscillation the air particles along its "back" or opposite surface. These vibrating air particles radiate as sound energy into the space on the back side of the wall. Thus, an incident sound wave excites the front side of the wall, and the wall re-radiates the sound wave from its back side. (If the wall is at all porous, some sound-oscillating air particles—can actually pass through the pores of the wall.)

Generally, a lightweight wall will be more easily excited by an incident sound wave than will a heavyweight wall and therefore will "transmit" more radiated energy to the other side. This generalization gives rise to the effect known as "the mass law" in acoustics. To a first approximation, "the mass law" suggests that for each doubling of the surface weight of the wall there will be about 5 or 6 dB less transmitted sound. The mass law also suggests that for each doubling of the frequency of the sound there will be about 5 or 6 dB less transmitted sound. Of course, there are many qualifications to these generalities. Details of these qualifications are not presented here; but the "transmission loss" data given in the tables reflect these effects.

The approximate "transmission loss" or "TL" values, expressed in dB, of a number of typical wall constructions are given in Tables 13-22 at the end of the text:

Table No. Construction Material

- 13 Solid, dense concrete or masonry
- 14 Hollow-core concrete or masonry
- 15 Stud-type partitions
- 16 Metal panel partition and Industrial acoustic doors
- 17 Glass walls or windows
- 18 Double-glass construction
- 19 Wood or plywood, including 2-inch thick solid wood door
- 20 Plaster
- 21 Aluminum
- 22 Steel

It is important to realize that the TL of a wall is merely the ratio, expressed in decibels, of the sound transmitted by a wall to the airborne sound incident upon the opposite side of the wall. Thus, the TL of a wall is a performance characteristic that is entirely a function of the wall weight and material. And its numerical value is not influenced by the acoustic environment on either side of the wall or the area of the wall.

#### 5.5 Noise Reduction of Walls ("NR")

The objective of this discussion on walls is to enable us to determine quantitatively the wall that is required to separate a noisy room from a quiet room. For example, if the noise level in an engine room is 90 dB at a particular octave frequency band and the noise level desired in an adjoining room is 40 dB in that same frequency band, then the wall must provide a "noise reduction" of 50 dB. But, the "noise reduction" provided by that wall involves more than merely the TL of the wall.

It is probably obvious that a wall with a relatively small area will transmit less total noise energy than will a wall with a relatively large area, even though each square foot of the wall has the same TL value. Also, it is probably obvious that the sound level in the "receiving room" will be influenced by the amount of acoustic absorption in the receiving room, that is, the SPL will be relatively high in a "live" receiving room having little or no acoustic absorption whereas it will be relatively low in a "dead" receiving room having large amounts of acoustic absorption. Thus, when noise travels through a wall from one room to an adjoining room, three factors are involved: (1) the TL of the wall (as in Tables 13-22); (2) the area of the wall that is transmitting the noise (that part of the wall area that is common to both the noisy room and the adjoining room); and (3) the acoustic characteristics of the adjoining room that receives the transmitted noise. The term "noise reduc-tion" of a wall (abbreviated "NR") is the term that includes all three of these factors. In practice, the area of the common transmitting wall and the acoustic characteristics of the receiving room can be incorporated into a single "wall correction term"<sup>2</sup> which in turn can be applied to the TL of that wall. Then,

<sup>2&</sup>quot;Wall correction term" is coined especially for use in this manual. It actually includes several factors combined into one term in the interest of simplifying the details.

the "noise reduction" provided by the wall separating the two rooms is the *the SPL difference between the two rooms.* 

However, the NR for any wall is equal to the TL of the wall plus the "wall correction term" (designated here as "C"),  $or^3$ 

#### NR = TL + C

In these two equations TL and C have known values for specific situations, and the SPL of the source room can be determined from Data Sheet 5. Then, for a given set of conditions, the SPL in the receiver room can be determined, or

#### SPL = SPL + TL + C receiver source room room

The approximate values of wall TL can be found in Tables 13-22, and the approximate values of "C" can be found in Table 23, where it is necessary to know the total receiving room surface area, the common wall area and the condition of acoustic treatment of the receiving room. In this simplified approach, several approximations are made, and the resulting value of "C" is used equally in all eight frequency bands.

#### 5.6 Example, Data Sheet 6

From all the material thus far presented in the manual, it is now possible to make a very significant calculation. Suppose it is desired to know if an 8-inch hollow-core dense concrete block wall will be adequate to separate an engine room on the top floor of a building from the bedroom of a penthouse apartment immediately beside the engine room.

Suppose the engine and engine room are the same as those used in Example 5.2, where the engine room reverberant and close-in SPLs were found in Items 9 and 10 of the sample Data Sheet 5. Suppose that the bedroom wall common to the engine room is 8 feet high and 12 feet long and suppose that the bedroom has a total inside surface area of 792 square feet and a 12-foot-x-15-foot ceiling of 1-inch-

thick acoustic absorption material. Also, the floor is carpeted.

Sample Data Sheet 6 indicates the steps required to estimate the SPL in the bedroom ("receiving room") due to the engine in the engine room ("transmitting room"). In filling in the blanks of Data Sheet 6, note that the footnote of Item 4 permits the addition of extra acoustic absorption to the receiving room due to the presence of carpet, drapes or upholstered furniture.

In a more thorough analysis, the foom absorption could be calculated in considerable detail; but here fairly simple estimates are considered adequate. Another judgement decision might be involved in filling in the blanks of Item 12 of Data Sheet 6. If the transmitting wall is quite close to the engine noise source, the noise levels at that wall will be higher than if the wall is more remote from the engine. Hence, for a close wall (say, under 5 feet from the engine), the close-in noise levels of the engine room should be used, i.e., Item 10 of Data Sheet 5; while for a greater distance (say, 5 feet or more from the engine), the reverberant noise levels of the engine room should be used, i.e., Item 9 of Data Sheet 5. In this particular example, assume that the wall is more than 5 feet from the engine and use the reverberant SPL.

Item 13 of sample Data Sheet 6 shows the following SPL values in the bedroom ("receiving room") for the octave frequency bands:

60 62 60 58 53 45 33 21 dB

The question asked by this example is whether the 8-inch-thick hollow-core concrete block wall will yield acceptable noise levels in the bedroom. Recall from the discussion of noise criteria (Paragraph 1.1) and from Table 1 at the end of the text that the noise criterion designation for sleeping (Category 1 in Table 1) is NC-20 to NC-30. From Table 2, the SPL values for NC-20 and NC-30 in the octave frequency bands are as follows:

NC-20: 51 40 33 26 22 19 17 16 NC-30: 57 48 41 35 31 29 28 27

A comparison of the estimated bedroom noise

<sup>&</sup>lt;sup>3</sup>Solution to NR = TL + C using the procedure in this manual results in a minus quantity. Caution: Observe algebraic signs in adding "noise" (+) to "reductions of noise" (-) in this manual; e.g., 10 + (-5) + (-2) = 3.

Data She								
Data Sheet 6						(Paragrap	(Paragraph 5.6 Exampl	
	50	und frans	mission to .	Aajonning Ki	iom inrough co	40000 **40		
Sound Transmitting Room						Sound R	Sound Receiving Roon	
Engine Room						Bedroom		
1. Area o	f Common V	Vall That Ti	ansmits Nois	a: <u>96</u>	sq, ft.			
2, Total I	nterior Surf	ace Area of	Receiving Ro	om (including	floor):79	2sq. ft.		
3, Total I {Item :	nterior Surf 3 ≈ item 2 ∻	ace Area Di Item 1)	vided by Arei	of Common \	Vall:8.2	-		
4. Area o	f Acoustic T	reatment in	Receiving R	270	)sq. ft.			
5, Per cer (Item !	nt Area of Ri 5 ≠ 100 X It	eceiving Ro em 4 ÷ Iterr	om Covered ( 2)	y Acoustic Tr	eatment <u>34</u>			
6. Thickr	less of Absor	ption Mate	rial: 3/4	in.—t in. 🖾	, 1-1/2 in.—2 in.	D		
7. Acoust Condit	ic Treatmen ion 1 🛛,	t Condition Conditi	of Receiving on 2 🖸,	Room from It Condition 3	ems 5 and 6, as Dei 凶, Condition	ined in Footnotes	of Table 23	
8. Wall Co	orrection Te	rm "C" fror	n Table 23 fo	r Item 3 and It	em 7 Conditions:_	+1	dB	
[				Frequency Ba	nd in Hz		·····	
63	125	250	500	1,000	2,000	4,000	8,000	
	ed Wall Con:	struction M	aterial: <u>8</u>	in. hollo	w-core dense	concrete bl	ock	
9. Expect		Wall from <sup>-</sup>	Tables 13-22	or Other Sourc	8			
9. Expect 10. Estimat	ted "IL" of						50	
9. Expect 10. Estima 31	33	35	36	41	48	54		
9. Expect 10. Estima 31 11. Estimat (Item 1 Caution	33 1 = 1 tem 10 1 = 0 tem 10 1; Observe al	35 Wall in Equ + Item B): gebraic sign	36 uation NR =	4고 TL + C, Using ng items!	48 Values from I tems	54 B and 10		
9. Expect 10. Estima 31 11. Estima (Item 1 Caution 32	33 ted "NR" of 1 = Item 10 1; Observe al 34	35 Wall in Equ + I tem B): gebroic sigr 36	36 Juation NR = In combinin 37	41 7L + C, Using 19 items! 42	48 Values from I tems 49	54 B and 10 55	60	
9. Expect 10. Estima 31 11. Estima (Item 1 Caution 32 12. Estimat ft, or M	ied "NR" of 1 = Item 10 1; Observe al 34 :ed SPL in Si ore from Co	35 Wall in Eq + Item B): gebraic sign 36 ound Trans	36 Jation NR = Is In combinin 37 Mitting Room	41 TL + C, Using items! 42 from Data Sh Close-in SPL If	48 Values from Items 49 eet 5 (Use Item 9 I Source Is Less Tha	54 B and 10 55 Reverberant SPL I n 5 ft. from Com	60 f Source Is 5 mon Wall}:	
9. Expect 10. Estima 31 11. Estima (Item 1 Caution 32 12. Estimat ft. or M 92	ted "IL" of 33 ted "NR" of 1 = 1 tem 10 1 : Observe al 34 ted SPL in Si ore from Co 96	35 Wall in Equ + Item B): Igebraic sign 36 ound Trans ommon Wall 96	36 Juation NR = Is In combinin 37 Mitting Room or Item 10 C 95	41 TL + C, Using ig items! 42 from Data Sh Close-in SPL If 95	48 Values from I tems 49 eet 5 (Use I tem 9 I Source Is Less Tha 94	54 B and 10 55 Reverberant SPL I n 5 ft. from Com	60 f Source Is 5 mon Wall): 81	
9. Expect 10. Estima 31 11. Estima (Item 1 Caution 32 12. Estimat ft, or M 92 13. Estimat	ted "IL" of 33 ted "NR" of 1 = Item 10 1: Observe al 34 ted SPL in Si ore from Co 96 ed SPL in R	35 Wall in Equ + Item B): gebraic sign 36 ound Trans ommon Wall 96 eceiving Ro	36 Juation NR = Is In combinin 37 mitting Room I or Item 10 ( 95 som (Item 13	41 TL + C, Using igitems! 42 h from Data Sh Close-in SPL If 95 = Item 12	48 Values from I tems 49 eet 5 (Use I tem 9 I Source Is Less Tha 94	54 B and 10 55 Reverberant SPL I n 5 ft. from Com 88	60 f Source Is 5 mon Wall}: 81	

levels with the NC-20 to NC-30 noise levels shows that the bedroom noise levels would be about 20 to 30 dB too high. If these noise levels were only a few decibels too high, this might be an almost acceptable condition (a discussion of "tolerances" appears later in Section 7), but a 20 to 30 dB noise excess would be completely unacceptable. Thus, the use of an 8-inch hollow-core concrete block wall would not be adequate to separate the bedroom from the engine room.

100

Actually, if this example were carried out again using a 12-inch-thick, solid dense concrete block wall, it would still be found inadequate to separate a bedroom from an engine room. This, then, essentially dictates that a sleeping area involving NC-20 to NC-30 noise criteria not be located immediately adjoining an engine room on the same floor. (Later comments are devoted to some special design requirements for (1) allowing a Category 3 or 4 area to adjoin immediately an engine room and (2) allowing a Category 1 or 2 area to be reasonably near an engine room.) Recall that the example used here was introduced primarily to illustrate the use of Data Sheet 6 for calculating the SPL in a room adjoining the engine room, but the example was also chosen to illustrate a potentially serious problem.

#### 5.7 Doors and Windows

It is probably fairly obvious that if a 12-inchthick, solid concrete block wall (at 144 lb/sq ft surface weight) were required to separate a noisy room from a quiet room, an ungasketed lightweight wood door or a single thickness of 1/4-in, glass window (each less than 4 lb/sq ft surface weight) would certainly represent a noticeable noise leakage path through that wall. If the noise reduction value of a wall is to be preserved, the doors and windows must be specially selected.

Where the area of a door or window is only a small part of the area of the wall in which it is installed, it is possible to permit the door or window to have a somewhat lower TL than the wall without significantly harming the total effectiveness of the wall. As examples, if the door or window area is 20% of the wall area, the TL of the door or window can be 3 dB lower than that of the wall; if 10%, 6 dB

lower; and if only 5%, the door or window TL can be 10 dB lower than the wall TL. For these conditions the total effectiveness of the wall would be decreased only 1 dB, which can be considered negligible for most practical installations. These and other combinations of window or door area and relative TL and the resulting loss of TL of the composite wall containing them are given in Table 24 (including an example). (This table can also be used to determine the effective TL of a wall made up of two different portions, where the two portions have different TLs, such as a 10-inchthick poured solid concrete wall having a knockout panel of 6-inch-thick concrete block.)

The approximate TL of a 2-inch, solid wood door, gasketed around all edges, is given in Table 19 (see Footnote 2 of Table 19), and the approximate TL of a 4-inch-thick and a 6-inch-thick industrial-type "acoustic door" is given in Table 16. The approximate TL of single thicknesses of glass is given in Table 17 and that of a few double-glass combinations is given in Table 18.

In some situations, the structural requirements may exceed the acoustical requirements of a wall, in which case the door or window can have a TL much lower than that of the wall. A few generalizations that should aid in the selection of a door or window that will be somewhat acoustically compatible with the wall, even though the door or window TL may not exactly meet the values suggested as a function of their area relative to the total wall area, are:

- Where the acoustic design requires a minimum, simple, single-wall construction, such as conventional stud partitions, movable metal partitions or 4-inch or 6-inch hollow-core concrete block, use ungasketed hollow-core wood doors or ungasketed metal panel doors and minimum 1/4-inch-thick glass windows.
- Where the acoustic design requires somewhat more than minimum wall construction (such as staggered stud construction, 4-inch or 6-inch solid core concrete or masonry, or acoustically filled metal panel partitions), use gasketed solid-core wood doors or minimum 1-3/4-inch hol-

Iow metal doors packed with dense mineral or glass fiber, or special 1-3/4-inch- to 2-inch-thick acoustic doors with gasketing, and use windows of minimum area made up of double panes of at least 1/4 inch-thick glass with at least 2-inch air space, or windows of larger but limited area made up of double panes of at least 1/4-inch-thick glass with 4-inch to 6-inch air space. For area details, check against Table 24 data.

- 3. Where stringent acoustic requirements must be met, adhere to the door or window TL requirements given in Table 24 as a function of per cent area of the total wall. Use special acoustic doors or provide "sound locks" with gasketed double doors, as in Item (2) above, such that doors are spaced at least 5 to 6 feet apart in an acoustically lined vestibule or corridor. Use double-glass windows with maximum possible air space and glass thickness and minimum practical area. For slight improvement, the panes may be tilted relative to one another, and the interior surfaces of the window framing can be given an acoustic lining.
- 4. Where doors are obvious leakage paths for unwanted noise, locate them in positions that will provide minimum disturbance or maximum distance from the important work area of the room, and provide acoustic absorption in the "receiving room."

#### 5.8 Transmission Loss of Floor-Ceiling Combinations

The transmission loss of a simple dense concrete floor slab alone is approximately the same as given in Table 13 for dense poured concrete. However, in most building situations, a ceiling of some type is supported below the floor slab.

Five different floor-ceiling combinations are considered here. Under no condition should a reciprocating or turbine engine be mounted on framed wood flooring or on typical lightweight metal deck with 2- to 3-inch-thick concrete surface. These floor constructions are not stiff enough or massive enough to support heavy machinery or to give an adequate base for an engine mounting system.

The five floor-ceiling combinations are discussed in the following paragraphs. (Note: All floor slabs are assumed to be of *dense concrete* [140-150 lb/cu ft density].)

Type 1 floor-ceiling combination is the concrete floor slab that has acoustic tiles cemented directly to the underside. It is important to realize that the acoustic tiles add nothing to the transmission loss of the floor slab. The acoustic tiles only provide acoustic absorption in the room in which they are located and hence provide a degree of noise reduction in the room, as shown by the calculations in Data Sheets 5 and 6. The estimated TL of a Type 1 floor-ceiling is given in Table 25 for a few typical floor slab thicknesses.

Type 2 floor-ceiling combination consists of a concrete floor slab, below which is suspended a typical low density acoustic tile ceiling in a mechanical support system. To qualify for the Type 2 combination the acoustic tile should be not less than 3/4 inch thick, and should have a noise reduction coefficient ("NRC") of at least 0.65 (when mounted as specified by the Acoustical Materials Association; see Footnote 1 in Paragraph 5.1). The air space between the suspended ceiling and the concrete slab above should be at least 15 inches, but the TL improves if the air space is larger than this. The estimated TL of a Type 2 floor-ceiling is given in Table 26 for a few typical dimensions of concrete floor slab thickness and air space.

Type 3 floor-ceiling combination is very similar to the Type 2 combination, except that the acoustic tile material is of the "high TL" variety. This means that the material is of high density and usually has a foil backing to decrease the porosity of the back surface of the material. (Ask the acoustic tile representative to identify his "high TL" material.) One possible version of the Type 3 combination includes the suspended ceiling system that consists of a lightweight metal panel sandwich construction consisting of a perforated panel on the lower surface and a solid panel on the upper surface, with acoustic absorption material in-between. The minimum NRC for the Type 3 acoustic material must be 0.65. The estimated TL of a Type 3 floor-ceiling is given in Table 27 for a few typical dimensions of concrete floor slab thickness and air space.

Type 4 floor-ceiling combination consists of a concrete floor slab, an air space, and a resiliently supported plaster ceiling. This combination is for use in critical situations where a high TL is required. The plaster ceiling should have at least 1 inch thickness of high density plaster (minimum 12 lb/sq ft surface weight), and the air space should be at least 18 inches thick.

The ceiling should be supported on resilient ceiling hangers that provide at least 1/10-inch static deflection under load. Neoprene-inshear or compressed glass fiber hangers can be used, or steel springs can be used if they include a pad or disc of neoprene or glass fiber in the mount. A thick felt pad hanger arrangement can be used if it meets the static deflection requirement. The hanger system must not have metal-to-metal short-circuit paths around the isolation material of the hanger.

Where the plaster ceiling meets the vertical wall surface, the perimeter edge of the ceiling must not make rigid contact with the wall member. A 1/4-inch open joint should be provided at this edge, which is filled with a non-hardening caulking, or mastic or fibrous packing after the ceiling plaster is set.

Provide State

の時来は、それにはないなけれるというため

Constant S

10.014

The estimated TL of a Type 4 floor-ceiling combination is given in Table 28 for a few typical dimensions of floor slab, air space and ceiling thicknesses. It is cautioned that this combination is for use in critical situations, and special care must be exercised to produce a good, resiliently supported, non-porous, dense ceiling. Acoustic tile can be added to the underside of the plaster ceiling but it will not change the transmission loss of the combination; it will only add to the acoustic absorption of the room.

Type 5 floor-ceiling combination is the same as the Type 4 combination, except that a "floating concrete floor" is mounted on top of the structural floor slab. The floating concrete floor should not support the engine assembly or any other large operating equipment. It should extend over all the engine

Total in the second of the second strategic the second s

room floor area within 20 feet of the engine assembly, but not under any vibration isolated concrete inertia bases carrying specific pieces of operating machinery. The floating concrete floor should be supported off the structure floor at a height of at least 2 inches with the use of properly spaced blocks of compressed glass fiber or multiple-layers of ribbed or waffle-pattern neoprene pads or steel springs (in series with two layers of ribbed or waffle-pattern neoprene pads).

The density and loading of the compressed glass fiber or neoprene pads should follow the manufacturers' recommendations. If steel springs are used, their static deflection should not be less than 1/4 inch. The 2-inch space between the floating slab and the structure slab should be covered with a 1-inch thickness of low-cost glass fiber or mineral wool blanket of 3 to 4 lb/cu ft density.

Around all the perimeter edges of the floating floor (around the walls and atound all concrete inertia bases within the floating floor area) there should be 1-inch gaps that are later packed with mastic or fibrous filling and then sealed with a waterproof non-hardening caulking or sealing material. It would be advisable to provide a curb arrangement around the perimeter of the floated slab to help discourage water leakage into the sealed perimeter joints, because, if the floor becomes flooded, water should not fill up the 2-inch space under the floated slab.

As a prevention against this occurrence, several floor drains should be set in the structure slab under the floating slab to provide run off of any water leakage into this cavity space.

As with the Type 4 combination, the Type 5 combination includes a resiliently supported plaster ceiling under the structure slab. The estimated TL of a Type 5 floor-ceiling combination is given in Table 29 for a few typical dimensions of floating floor slab in combination with the Type 4 structures of Table 28. The Type 5 combination may be required for an engine room floor in certain critical situations but it probably would not be required in any other applications within the scope of this manual. It is to be noted that the floating slab is intended to improve the airborne TL of a floor; it is not suggested here as a vibration

isolation mounting base for large equipment, although it will provide certain benefits to some structure-borne noise of pipe supports, duct supports, drainage lines, electrical conduit and the like. The floating slab is not designed here to support pumps, fans, compressors, engines, motors, refrigeration equipment, and the like.

As a general rule, to be reinforced later in the section under vibration isolation, the engineroom structural-floor slab for an upper floor in a multi-floor building should not be less than 6 inches thick for a gas turbine engine used to drive a completely rotary-action device, or less than 8 inches thick for a gas turbine engine used to drive a reciprocatingaction device. For a reciprocating engine drive, the structural floor slab should not be less than 10 inches thick. These suggestions are based on acoustic considerations only and are not intended to represent structural requirements of the building. Even thicker floor slabs will be slightly more beneficial acoustically. Where possible, the engine assembly should be located over principal or secondary beams in the flooring layout.

In the upper frequency bands of Tables 25-29, extremely high TL values (say, anything above 60 or 65 dB) are indicated as possible. In practice, these values cannot be achieved without making a concentrated effort to stop all escape paths of airborne and structure-borne noise.

5.9 Noise Reduction of Floor-Ceiling Combinations

Paragraph 5.5 discussed the conversion of transmission loss of a wall into the noise reduction of a wall by use of the "wall correction term," designated by the letter "C" in Paragraph 5.5 and in Table 23. The same type of correction must be applied to convert the TL of a floor-ceiling combination to its NR value. This applies, of course, to the situation in which the engine room is immediately above or below an adjoining area of concern. For identification purposes, the term is called "floor correction term" here, but it is represented by the same letter "C." It is also obtained from Table 23, based on (1) room absorption; (2) total interior area; and (3) com-

mon floor-ceiling area of the receiving room. The value of "C" will differ, of course, from room-to-room so it must be redetermined for each room of interest above, below or beside an engine room.

#### 5.10 Example, Data Sheet 7

Suppose an on-site total energy plant is considered for location on the top floor of an office building. Suppose three 450-hp, 1,800-rpm reciprocating natural gas engines would be located in the engine room directly over a suite of private offices. Suppose a typical office is 10 feet x 15 feet x 8 feet and has a carpet and drapes but no acoustic tile ceiling. Because of the very critical location, suppose a Type 5 floor-ceiling is first calculated to determine the feasibility of this location.

Sample Data Sheet 7 indicates the steps to be taken in the calculation. Before Data Sheet 7 can be completed it is necessary to determine the engine room SPL from Data Sheets 1 and 5. These sample calculations are not given here, but for an assumed engine room of dimensions of 40 feet x 80 feet x 15 feet with a full 50% of all interior surfaces covered with 2-inch acoustic absorption material, the following engine room SPLs can be calculated:

- (a) The reverberant SPL due to one engine would be 87 91 91 90 90 89 83 76 dB
- (b) The close-in SPL due to one engine would be
  - 90 94 94 93 93 92 86 79 dB
- (c) The reverberant SPL due to three engines would be 92 96 96 95 95 94 88 81 dB

Since the floor of concern is immediately under the engines, the close-in SPL of each single engine would normally apply. But in this room, the reverberant SPL due to all three engines is slightly higher than the close-in SPL of each engine, so the reverberant SPL for all three engines is used, i.e.,

92 96 96 95 95 94 88 81 dB

Suppose the Type 5 floor-ceiling consists of a
30	und Tran Engl	smitting Re Line Room	oom 1	·			Sound Prive	Receiving Room ate Office
-		(			The Blatance	150		
1	, Area oi	common r	noo Area of I	Finat Fransir Resolution Ro	am finaludina :		) 	
2	Tenet le	terior surfa	ace Area of		om (including)	1100r):	\$4, 11.	
3	(Item 3	= Item 2 ÷	Item 1)	4.7	or Common P	ioor-ceiling;		
4	. Area of	Acoustic T	reatment in	Receiving Ro	00m*: <u>75</u>	sq. f	t.	
5	. Percent (Item Б	Area of Re = 100 X Ite	eceiving Roc em 4 ÷ Item	om Covered b 2)1	y Acoustic Tre	atment		
6	. Thickne	ss of Absor	ption Mater	ial: 3/4 in	-1 in. £] 1	-1/2 in.—2 in. 🖾		
7	. Acoustic	Treatmen	t Condition	of Receiving	Room from Ite	ems 5 and 6, as D	efined in Footnot	tes of Table 23:
p	Elect C:			11 ≠ L21, m. Table 22 4		, conditio	an+⊔ 	dB
υ,					Frequency Ban	d in Hz	3, <u></u>	
	63	125	250	500	1,000	2,000	4,000	8,000
0.	Type 1 [ Approxi (Interpol of Table:	Ty: mate "TL" ate Betwee \$ 25-28}:	of Floor-Ce of Floor-Ce on TL Value	Type 3 [] Biling Type of Shown, If D	Type 4 🛛 Item 9 for Nea Jesired, But Do	Type 5 D arest Applicable Not Exceed TL	Dimensions, from Values Shown in I	Tables 25-29 Right-Hand Column
10.	Type 1 [ Approxil (Interpol of Table: 50	Ty; mate "TL" ato Betwee 25-28): 54	of Floor-Ce of Floor-Ce n TL Value 58	Type 3 [] Hiling Type of Shown, If D 64	Type 4 D Item 9 for Nez Pesired, But Do	Type 5 D arest Applicable Not Exceed TL 78	Dimensions, from Values Shown in I 82	Tables 25-29 Right-Hand Column 88
10.	Type 1 I Approximate (Interpolition of Tables) 50 Estimate (Item 11 Caution:	Ty; mate "TL" (ato Betwee s 25-28): 54 d "NR" of = Item 10 Observe al	of Floor-Ce of Floor-Ce TL Value 58 Floor-Cellin + Item 8): gebraic sign	Type 3 [] illing Type of s Shown, If D 64 ng in Equation s In combinin	Type 4 □ Item 9 for Nea Jesired, But Do 72 n NR = TL + C, g itemsi	Type 5 D arest Applicable Not Exceed TL 78 Using Values of	Dimensions, from Values Shown in 1 82 I tems 8 and 10	Tables 25-29 Right-Hand Column 88
10.	Type 1 E Approximate (Interpol of Tables) 50 Estimate (Item 11 Caution: 46	Ty; mate "TL" ate Betwee 25-28): 51 d "NR" of = item 10 Observe al 50	of Floor-Ce of Floor-Ce n TL Value 58 Floor-Cellin + Item 8): gebraic sign 54	Type 3 [] Hiling Type of s Shown, If D 64 Ing in Equation s In combinin 60	Type 4 □ Item 9 for Nez lesired, But Do 72 n NR = TL + C, g itemst 68	Type 5 Dt arest Applicable Not Exceed TL 78 Using Values of 74	Dimensions, from Values Shown in 1 82 Herms 8 and 10 78	Tables 25-29 Right-Hand Column 88 88
10.  1.	Type 1 C Approximate (Interpol of Tables) 50 Estimate (Item 11 Caution: 46 Estimate Above th	Typ mate "TL" ate Betwee \$25-28}: 51 d "NR" of = Item 10 Observe al 50 d SPL in So te Engine o	se 2 of Floor-Ce rn TL Value 58 Floor-Cellin + Item 8): gebraic sign 54 Sund Transm r Item 10 C	Type 3 [] Siling Type of s Shown, If D 64 Ing in Equation s In combinin 60 Initting Room lose-in SPL fo	Type 4 □ Item 9 for Nez lesired, But Do 72 n NR = TL + C, g itemst 68 from Data She or Floor Under	Type 5 D arest Applicable Not Exceed TL 78 Using Values of 74 ret 5 (Use Item 9 Engine):	Dimensions, from Values Shown in 1 82 I Items 8 and 10 78 Reverberant SPL	Tables 25-29 Right-Hand Column 88 88 84 for Floor-Ceiling
10.	Type 1 C Approximate (Interpol of Tables) 50 Estimate (Item 11 Caution: 46 Estimate Above th 92	Typ mate "TL" ate Betwee 25-28): 51 d "NR" of # Item 10 Observe al 50 d SPL in So te Engine o 96	of Floor-Ce of Floor-Ce r TL Value 58 Floor-Cellin + Item 8): gebraic sign 54 54 ound Transm r Item 10 C 96	Type 3 [] Hiling Type of s Shown, If D 64 Ing in Equation s In combinin 60 Initting Room lose-in SPL fo 95	Type 4 □ Item 9 for Nez lesired, But Do 72 n NR = TL + C, g itemst 68 from Data She or Floor Under 95	Type 5 D arest Applicable Not Exceed TL 78 Using Values of 74 et 5 (Use Item 9 Engine): 94	Dimensions, from Values Shown in 82 I Items 8 and 10 78 Reverberant SPL 88	Tables 25-29 Right-Hand Column 88 88 60 Floor-Ceiling 81
10.   1.   2.	Type 1 C Approximation (Interpol of Tables) 50 Estimate (Item 11 Caution: 46 Estimate Above th 92 Estimate	Typ mate "TL" ato Betwee \$ 25-28}: 54 d "NR" of = I tem 10 Observe al 50 d SPL in St 96 d SPL in R	se 2 of Floor-Ce en TL Value 58 Floor-Cellin + Item 8): gebraic sign 54 54 54 ound Transn r Item 10 C 96 ecciving Roc	Type 3 [] Filing Type of s Shown, If D 64 ng in Equation s in combinin 60 nitting Room lose-in SPL fo 95 om (Item 13	Type 4 $\Box$ Item 9 for Nez lesired, But Do 72 n NR = 7L + C, g items! 68 from Data She or Floor Under 95 = Item 12 - Ite	Type 5 Dt arest Applicable Not Exceed TL 78 Using Values of 74 et 5 (Use Item 9 Engine): 94 m 11)	Dimensions, from Values Shown in 82 I tems 8 and 10 78 Reverberant SPL 88	Tables 25-29 Right-Hand Column 88 88 84 for Floor-Ceiling 81
10. 11. 12.	Type 1 C Approximate (Interpol of Tables) 50 Estimate (Item 11 Caution: 46 Estimate Above th 92 Estimate 46	Typ mate "TL" ate Betwee 25-28): 54 d "NR" of 1 tem 10 Observe al 50 d SPL in So e Engine o 96 d SPL in R 46	se 2 of Floor-Ce rn TL Value 58 Floor-Cellin + Item 8): gebraic sign 54 sound Transm r Item 10 C 96 ecciving Roc 42	Type 3 [] Filing Type of s Shown, If D 64 Ing in Equation s In combinin 60 Initting Room lose-in SPL fo 95 om (Item 13 35	Type 4 □         Item 9 for Nez         lesired, But Do         72         n NR = 7L + C,         g itemst         68         from Data She         or Floor Under         95         = Item 12 - Ite         27	Type 5 D arest Applicable Not Exceed TL 78 Using Values of 74 et 5 (Use Item 9 Engine): 94 m 11) 20	Dimensions, from Values Shown in 1 82 I Items 8 and 10 78 Reverberant SPL 88	Tables 25-29 Right-Hand Column 88 88 60 Floor-Ceiling 81 -3

n 1997 - Santa 1997 - Santa Sa

....

¥

**于是国家的,如果实现是有的情况。在这些情况和这些情况,我们就是我们就可能能是我们的是不能的。""你们,你是你们不能的问题。"** 

.

ι.,

1

4-inch-thick floating concrete slab supported resiliently 2 inches above a 10-inch-thick dense concrete structure floor slab, with a 30-inch air space to a 1-1/2-inch-thick dense plaster ceiling supported resiliently below the slab. Tables 28 and 29 provide the TL estimate for this combination.

Item 13 of sample Data Sheet 7 yields the following estimated SPL values in the typical office immediately under an engine

46 46 42 35 27 20 10 -3 dBThese levels may be compared with the applicable NC-30 to NC-35 criterion levels for a private office (Tables 1 and 2) as follows.

NC-30: 57 48 41 35 31 29 28 27 dB NC-35: 60 52 45 40 36 34 33 32 dB

It is seen that the NC-30 criterion would be exceeded by 1 dB in only one frequency band, and the NC-35 criterion would be met in all frequency bands. This would probably be an acceptable solution, especially since the office ventilation system air supply would probably also produce some masking noise in the NC-30 to NC-35 range.

5.11 Special Situations

In the example shown in Paragraph 5.10 it is



Refrigeration system assembly is installed rigidly on massive concrete inertia block. Even when there is no nearby acoustically critical area, this type of mounting will reduce vibration of system components and piping, floor slab and building structure itself. (Photograph is courtesy of Caterpillar Tractor Company.) seen that a rather special floor-ceiling combination is required to separate a fairly noisy engine room from a fairly quiet private office. Even so, this particular Type 5 floor-ceiling combination would represent only a marginal solution if the same engine room were *directly over* a bedroom in an apartment building layout (and probably should not even be attempted without extra acoustical assistance). This raises the question: Can an engine room be located, say, on the top floor of an apartment building? The answer to this question is "yes" for a number of situations and "no" for a few situations. These are labeled "special situations" here and are discussed briefly.

- A. Reciprocating Engines:
  - Primarily for vibration considerations, reciprocating engines above about 1,000 hp each or below about 900 rpm should not be located on an upper floor of a building that will house occupants falling into Categories 1, 2 and 3 of Table 1, without the aid of an acoustical consultant to assure that certain special details are met. Vibration isolation is discussed later in the manual.
  - 2. Reciprocating engines above about 50 to 100 hp probably should not be located *directly above* a Category 1 or 2 area of Table 1, without the aid of an acoustical consultant to check for the required details.
  - 3. Reciprocating engines below about 1,000 hp each and above about 900 rpm can be located on an upper-level engine room floor (with proper vibration isolation, as discussed in Section 8) provided the location is horizontally removed from an NC-20 to NC-25 Category 1 or 2 area by at least 40 feet or two column lines of the building (whichever is the greater distance), or is horizontally removed from an NC-25 to NC-30 Category 1 or 2 area by at least 20 feet or one column line (whichever is the greater). These generalizations apply to a critical area on the floor below or be-

side the engine room, and there must be at least one good full-height wall (having a TL of greater than 25 dB in the 250, 500 and 1,000 Hz frequency bands) between the critical area and the engine room in addition to the walls and floors that enclose the engine room. This second wall must be located at least 10 feet beyond the enclosing wall of the engine room, preferably as much as 20 to 30 feet, if the building layout permits. These same suggestions may also be applied to a critical area located on the floor above an engine room floor but the horizontal separation distances can be relaxed slightly (if necessary), if all piping associated with the engine assembly are well isolated in and near the engine room.

- 4. Reciprocating engines of any size and speed can be located in a basement or on-grade position of any building as long as proper vibration isolation and suitable airborne noise control are provided. If a critical Category 1 or 2 area is located on the floor above the engine room floor, the horizontal separations discussed above are advised.
- 5. There need be no horizontal separation between the engine room and any critical area that is two floors, above or below the engine room floor, so long as proper vibration isolation and adequate airborne noise control are provided for the engine.
- B. Turbine Engines:

2001

 In terms of vibration considerations alone, there appears to be no size limit for a turbine engine located on an upper floor of a multi-floor building that houses occupants in Categories

 2 and 3 of Table 1, provided the engine drives entirely rotary-action devices (no reciprocating-action devices) and provided the engine assembly is properly vibration isolated. As a practical matter, the scope of this manual limits the size to about 5,000 kw.

 2. In terms of noise considerations alone, a turbine engine rated above about 500 kw probably will either require a Type 4 or Type 5 enclosure (as described briefly in the footnotes of Table 6) or horizontal separation, such as discussed above in Item 3 for reciprocating engines, if an NC-20 to NC-30 critical area is located near the engine room in an upper floor of a building. In addition, of course, proper vibration isolation and airborne sound control for the areas immediately surrounding the engine room should be provided, in accordance with Data Sheets 6 and 7, as applicable.

These special situations will become apparent when calculation of Data Sheet 6 or 7 reveals that a particular high-TL wall or floor-ceiling cannot produce low enough noise levels in the adjoining space to meet the noise criterion desired for that space. In effect, the need for horizontal separation, as discussed in the "special situations" above, is evidence of need for "double-wall construction" in order to attain a high TL value. The recommendation of a second wall at least 10 feet horizontally from the engine room wall would meet the "double-wall" requirement. There are more sophisticated double-wall structures that do



Solar gas turbine generator sets produce power and exhaust heat for chemical plant. Exhaust is ducted out of room. Operators have glass-walled control room to reduce noise. (Photograph is courtesy of Solar Division, International Harvester Company.)

NAMED CONTRACTOR OF THE OFFICE OFFICE OFFICE OFFICE OFFICE OFFICE OFFICE OFFICE OF THE OFFICE OFF

not require the 10 foot separation but they become rather complex structures and require special detailing and special attention during construction. If the building layout can accommodate the 10-foot horizontal separation of walls (on the same or immediately adjoining floors), no special wall details would be required (other than that the second wall be of such construction as to provide a TL of at least 25 dB in the three frequency bands listed above-see Tables 13-22 for possibilities).

### 5.12 Engine Room Precautions

Although there may be no obligation to consider the following factors in designing an engine room, there is a growing concern for the well-being of operating personnel and for operational efficiency. Two points are mentioned here regarding engine room occupancy by personnel.

First, conversation in the engine room is usually quite difficult. For this reason, audio communication should not be encouraged: do not install a loud bell to call the operator (use a flashing light instead) and do not install a telephone in the engine room (install the telephone in the control room).

Second, engine room noise levels are sufficiently high that an operator should not be required or expected to remain in the engine room during long periods without ear muffs or ear plugs. To overcome this possibility, a small control room or engineer's office should be provided beside the engine room. Operating personnel should use the control room except when required to be in the engine room.

Detailed analyses of allowable time intervals in engine rooms without the use of ear protectors have been worked out in the manual identified under Footnote 1 at the beginning of Section 3. The details are too lengthy to repeat here, but it would be a fair summary to state that operators should wear ear protectors inside an engine room with reciprocating engines every time they are likely to remain in the engine room for more than 10 minutes at a time. For intervals of less than 10 minutes, separated by at least 30-minute periods in a much quieter environment, ear protectors probably need not be worn. Probably, car protectors should be worn for all time intervals in an engine room containing turbine engines. To arrive at more specific recommendations, the reader should know the actual engine room SPLs and check them against the latest data of the "CHABA" committee4.



<sup>4</sup>"Hazardous Exposure to Intermittent and Steady-State Noise," National Academy of Science and National Research Council, Committee on Hearing, Bioacoustics and Biomechanics ("CIABA"), January 1965. (Also published in the Journal of the Acoustical Society of America, Vol. 39, No. 3, pp 451-464, March 1966.)





Ear protection is recommended for personnel exposed to high levels in engine room. (Photograph is courtesy of Northern Illinois Gas Company.)

## CONTROL OF AIRBORNE SOUND OUT-OF-DOORS

Section 5 was concerned mostly with the distribution and control of engine casing noise within the building served by the engine. In Section 6, primary concern is for the distribution and control of that portion of the engine noise that is radiated out-of-doors and that might be heard by the neighbors. The term "neighbor" is used here to designate the nearest person or property in a given direction to be protected against excessive noise from the engine. The neighbor may be a nearby resident or a nearby building or, in some cases, part of one's own building or its occupants.

### 6.1 Sound Pressure Level Out-of-Doors

When a sound source is free to radiate out-ofdoors, the sound usually travels out in all directions and becomes less intense at increasing distances from the source. For an "omnidirectional" or "non-directional" sound source (i.e., it radiates equally in all directions), the outdoor SPL at any distance, "D", from a source. PWL, is given by the relationship:

 $SPL = PWL - 10 \log (2\pi D^2) + 10 - \alpha D/1000$ 

where  $\alpha$  is the loss per 1,000 feet due to "molecular absorption" of sound in the air. For the manual, the latter terms of this equation have been combined into one term, called an "outdoor distance term," such that

SPL = PWL - Outdoor Distance Term

The "outdoor distance term" has been calculated and summarized in Table 30 for a range of distances from 10 feet to 7,200 feet. This assumes average atmospheric conditions and no sound barrier or obstruction between the source and the receiver. Weather conditions can influence sound transmission, particularly at the large distances, but usually the weather is sufficiently variable that it cannot be relied upon to give a permanent benefit to a noise problem. Hence, in this manual, average sound propagation conditions are assumed.

### 6.2 Example, Data Sheet 8

As an example of the use of Table 30, suppose it is desired to determine the SPL at a distance of 1,600 feet from the reciprocating engine unmuffled exhaust noise of Example 3.4, as shown by Item 7 of sample Data Sheet 2 for that example, assuming that the noise radiates uniformly in all directions from the exhaust pipe opening.

Sample Data Sheet 8 indicates the steps to be taken for obtaining the outdoor SPL at the 1,600-foot distance.

#### 6.3 Noise Reduction Provided by a Building

An intruding noise coming from an outdoor noise source may be heard by a neighbor who is either indoors in his own building or outdoors on his property. If he is outdoors he may judge the intruding noise against the more-or-less steady background noise due to other noises in the area. If he is indoors, he may tend to judge the noise by whether it is audible or identifiable or intrusive into his surroundings. If the noise, when heard indoors in the neighboring building, can be made to be no greater than the criterion levels of Table 2 for the Table 1 activity areas, it is quite likely that there will be no complaint against the noise.

When outdoor noise passes into a building, it suffers some noise reduction, even if the building has open windows. The actual amount of noise reduction (NR) depends on building construction, orientation, wall area, window area, open window area, interior ac-

. I THE TAKEN BUTCHENGEN AND THE REPORT OF THE

Data Sheet 8

(Paragraph 6.2 Example)

5. St. 19

### Estimated Outdoor Sound Pressure Level (SPL) Due to an Outdoor Sound Source PWL

1. Distance to Noise Source: 1,600 ft.

				Frequency Ban	d in Hz		·····
63	125	250	500	1,000	2,000	4,000	8,000

2. Total PWL of All Outdoor Noise Sources at Source Position:

- 1		· · · · · · · · · · · · · · · · · · ·	r			···		
	121	127	123	115	111	105	95	87
			L				L	·

3. Outdoor Distance Term from Table 30 for Item 1 Distance:

·			~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~				
62	62	62	63	64	67	75	85

4. Approximate Outdoor SPL at Distance of Item 1: (Item 4 = Item 2) Caution: Observe algebraic signs in combining items!

59	65	61	52	47	38	20	2
L	L	<u> </u>	L	L			

oustic absorption, etc. For practical purposes, however, the average NR values of Table 31 can be used in the manual.

To estimate the indoor SPL of a noise coming inside the building from outdoors, it is merely necessary to reduce the outdoor SPL by the amount of NR found in Table 31 for the appropriate wall and window condition. If the building has no windows at all, an approximate NR value can be estimated for the wall from the TL values of Tables 13-22. When the indoor SPL is estimated, it can be compared with the noise criterion levels of Tables 1 and 2 considered appropriate for the situation. If the indoor SPL is equal to or less than the criterion levels, there will generally be no noise problem. If the indoor SPL exceeds the criterion levels by more than a few decibels in one or more of the frequency bands, there may be a problem if no further noise reduction is applied.

### 6.4 Noise Reduction Provided by a Solid Barrier

ł

Frank Street

A solid wall, a building, a large mound of earth, a hill or some other type of solid structure, if large enough, can serve as a partial barrier to sound and can provide some noise reduction for a receiver located within the "shadow" produced by the barrier.

Table 32 shows the important dimensions that influence the amount of noise reduction provided by a barrier. For a barrier to be effective, it should extend as far as possible beyond the line-of-sight between any part of the noise source and any part of the receiver in both the vertical and horizontal directions. This distance beyond the line-of-sight is represented by "H" in the simple sketch included with Table 32.

Also there must be no nearby large reflecting surfaces that can reflect sound around the barrier into the shadow zone. The distance D in the sketch of Table 32 must be large compared to the distance R and height (or width) H. The attenuation values given in Table 32 will apply equally for the two conditions: (1) sound source at Point A and receiver at Point B, or (2) sound source at Point B and receiver at Point A.

The barrier loses effectiveness at very large distances because sound that passes over the top of the barrier may be bent back down to the ground by wind and temperature gradients. If D is greater than 1 mile, the attenuation values used should be only about onehalf the values given in Table 32. This is usually not a very important factor in a problem, however, because at a one-mile distance, the noise has become so reduced by the distance effect that the loss of effectiveness of the barrier is hardly noticed.

If a barrier wall is to be built or used as a noise control device, the TL of the wall (or building) should exceed by at least 10 dB in all frequency bands the excess attenuation to be expected from the wall.

If the barrier is a large "thick" building, the distance R should be taken from Point A to the near wall of the building, and the height H should be the height of the building at that near wall. There should be no large openings entirely through the building that would destroy the effectiveness of the building as a barrier. A few small open windows in the near and far walls would probably be acceptable, provided the interior rooms are large.

It should be noted that a large reflecting surface, such as the barrier wall, may reflect more sound in the opposite direction than there would have been with no wall at all present. If there is no special focussing effect, the wall may produce about 2 or 3 dB higher levels in the direction of the reflected sound.

### 6.5 Noise Reduction Provided by Dense Woods

Heavy dense growths of woods provide a small amount of sound attenuation. To be effective both winter and summer, there should be a reasonable mixture of both deciduous and evergreen trees. Also the ground cover should be sufficiently dense that sound cannot pass through the lower, less dense parts of the tree growth. For dense woods of several hundred feet depth, the sound may pass over the tops of the trees, in which case the attenuation through the trees should never be considered greater than the excess attenuation over the trees, as determined from the ap-

uinen mari erranten ander den de arreiten fin het fin het in de sener en de sener en de sener en de sener en d I

plication of Table 32, treating the woods as though it were a solid barrier.

Table 33 gives the approximate noise reduction through dense woods, where dense woods are taken as having an average "visibility penetration" of about 70 to 100 feet. Occasional trees and hedges give no significant attenuation. "Visibility penetration" is the average maximum distance in the woods at which some small portions of a large (3-ft square) white cloth still can be seen. Footnotes in Table 33 pertain to reduced noise reduction rates for less dense and for shorter growths of trees.

### 6.6 Directivity Effect of a Stack Opening

If noise is emitted to the outdoors from the open end of a large duct, there would be more noise directly in front of the duct opening than there would be at the side of the duct. Table 34 gives some values to this "directivity effect" of a large ducted opening, assuming in this case it applies to the hot, high speed gases and noise emitted by the vertical exhaust stack of a large turbine engine. At positions directly in line with the stack opening  $(0^{\circ} \text{ in })$ 



Air inlet filter (with louvered face) and exhaust muffler for large reciprocating engine. (Photograph is courtesy of Diesel and Gas Turbine Progress.) Table 34), noise levels are higher than if this were a non-directional unducted source free to radiate equally in all directions. At positions to the side  $(90^{\circ} \text{ and } 135^{\circ})$ , the noise levels are lower than if this were a non-directional source. This effect is quite pronounced for large openings and tapers off to a negligible effect for small openings, where the duct size is not very large compared to the wavelength of the sound.

Field tests have shown that the same general effect holds for a large air *intake opening* that emits noise, with the exception that the effect at  $90^{\circ}$  and  $135^{\circ}$  is slightly greater, as shown by the footnote in Table 34.

If the inlet or exhaust duct is oriented horizontally and does not form a vertical stack, the general directivity effect of Table 34 can still be assumed, always measuring the angle from the axis of the opening, provided there are no nearby structures than can reflect higher sound levels into the shadow zone of the 90° and 135° regions.

Field data upon which Table 34 is based are quite sketchy and not very definitive. Thus, the actual directivity effects seem to be quite variable depending upon the geometrical variables of the several stacks tested. Some effect is known to exist, however, and reasonable averages are offered in Table 34.

Note that these data are based on vertical stacks that project at least 10 to 20 feet or more above other portions of the building and usually 20 to 40 feet above the ground. If such a stack now is considered as lying horizontally along the ground, the ground and any projections or structures on the ground in front of the stack opening can distort the directivity effect considerably.

If a large duct opening is located in the side wall of a large building, the building itself serves as a barrier that enhances the directivity effect. Less noise will be radiated in a direction of 180° from the axis of the duct, but all the geometric factors make it difficult to give general rules on "how much less." The values will fall somewhere between those of Tables 32 and 34, but this requires some judgement in the use of Table 32 for such a situation.

### 6.7 Noise Reduction Provided by "Reactive Mufflers"

"Reactive mufflers" are used almost entirely for gas and diesel reciprocating engine exhausts. Reactive mufflers usually consist of two or three large-volume chambers containing an internal labyrinth-like arrangement of baffles, compartments and perforated or slotted tubes. Sometimes also called "snubbers," these mufflers smooth out the flow of impulsive-type exhaust discharge and, by the arrangement of the internal components, attempt to reflect some sound energy back toward the source. These mufflers usually have no acoustic absorption material, although some manufacturers produce some models that contain an inside lining of hightemperature material. Many manufacturers produce a low-pressure-drop and a highpressure-drop line of mufflers, and each line comes in three different classes of noise reduction. The low-pressure-drop line is advertised largely for turbocharged engines, but the high-pressure-drop line can also be used with turbocharged engines, if the engine manufacturer approves the general exhaust layout. The high-pressure-drop line provides greater noise reduction. Typically, the three different classes of noise reduction are indicated by labels that somewhat relate to the degree of criticalness of the potential noise problem, such as "commercial," "standard," and "residential," and "industrial," and "semi-critical," and "critical," or similar series of names and models.

Table 35 gives approximate noise reduction values for the three classes of mufflers in the low- and high-pressure-drop lines. The three classes are identified here by relative size: small, medium and large. The small size has the lowest noise reduction; the large size has the highest noise reduction.

ļ

The engine manufacturer may recommend a maximum length and minimum diameter exhaust pipe for his engine, as these influence the back pressure applied to the engine exhaust. Exhaust pipe layout and location of the muffler also influence the noise reduction provided by the muffler.

Muffler manufacturers stress that the muffler

should be located as close as possible to the engine in order to avoid exhaust pipe resonances that coincide with engine firing rates. This is not always possible or practical; but it is well to know that if such a resonance should occur and the pipe vibrates strongly or the muffler seems to be ineffective, it may be possible to eliminate or reduce the pipe resonance by changing the pipe length 5 to 8 feet in between the engine and the muffler. For given engine configurations, some critical pipe lengths can be calculated but there is no assurance that resonance will or will not occur for those lengths, so the calculations are not included here.

### 6.8 Noise Reduction Provided by "Dissipative Mufflers"

"Dissipative mufflers" are made up of various arrangements of acoustically absorbent material that actually absorbs sound energy from the moving air or exhaust stream.

Dissipative mufflers are used in the air intake and gas exhaust of turbine engines. The most popular configuration is an array of "parallel



Twin set of gas turbine engines used to generate electrical power. Air intake to engines enters large mulfiled ducts at each side of engine housing (nearest air inlet duct shown with open access door in foreground.) Engine exhaust gases and noise pass through vertical mulfilers in this outdoor installation. (Photograph is courtesy of Diesel and Gas Turbine Progress.) baffles" placed in the air stream. The baffles may range from 2 inches to 16 inches thick, filled with glass fiber or mineral wool of such type as to stand up under the operating temperature (possibly 900° to 1,200°F in engine exhausts), with adequate internal construction and surface protection to resist the destruction and erosion by high-speed, turbulent flow. These baffles have been proved with many years' use in jet engine test cells. An experienced reputable manufacturer should be selected for these mufflers in order to insure proper quality of materials, design and workmanship and ultimately long-life and durability of the installation.

Where large amounts of attenuation are required in relatively short dimensions, it is necessary to install baffles at relatively close spacing, such that the total open area for air flow may be 30% to 70% of the total crosssection area of the stack or opening. This, in turn, produces pressure drop in the flow, so it is necessary to reach a compromise of cost, area, length and pressure drop in the final choice of muffler arrangement.

In general, thin parallel baffles (2 inches to 4 inches thick) are more effective in the higher frequencies, and thick parallel baffles (8 inches to 16 inches thick) are more effective in the lower frequencies. Unfortunately there are no baffles that are highly effective at very low frequencies, although it is known that large amounts of low frequency attenuation require small percentage open areas of muf-flers (such as 30% to 40%).

The field of muffler design and construction is so specialized that an architect or mechanical engineer should not undertake the design of these mufflers. Instead, it is suggested that noise reduction requirements for muffler application be worked out by use of the manual and in conjunction with the turbine engine manufacturer, and then be specified to the engine installer, contractor or muffler manufacturer. Tables 36 and 37 give the approximate noise reduction provided by some typical parallel-baffle mufflers.

For air intakes into gas turbine engines, the noise reduction requirement may be set by any one of the frequency bands; for turbine engine exhausts, the noise reduction requirement will probably be set by one of the low frequency bands.

In some critical installations it has been found that high flow speed through the muffler can produce noticeable "self-noise" that is itself a cause of concern. There is no exact schedule of air speed vs self-noise, but it has been found that a flow speed of about 175 to 200 feet per second is probably acceptable for engines located in fairly noisy commercial or industrial areas and of about 125 to 150 feet per second probably acceptable for engines located in or near quiet residential areas. Remember that, at engine exhaust temperatures, hot exhaust gas is of much lower density and has a higher total volume flow than the cool air intake into the engine.

### 6.9 Example, Data Sheet 9

The various noise reduction components discussed briefly under Paragraphs 6.3-6.8 can be summarized by use of an example and Data Sheet 9.

Suppose the example of Paragraph 6.2 is continued here. The unmuffled exhaust noise of a 200-hp turbocharged reciprocating engine was found to produce an outdoor SPL at 1,600 feet distance of:

59 65 61 52 47 38 20 2 dB

These values become Item 1 of sample Data Sheet 9. Suppose the neighboring building at 1,600 feet distance is a suburban residence having normally open windows. There is no major obstacle between the engine and the residence that is large enough to serve as a useful sound barrier, but there is a 200-footwide densely wooded area separating the engine from the residential area. The exhaust pipe opening is too small to have any significant directivity effect. Items 2-7 of Data Sheet 9 summarize the effects of these noise reduction contributions. The tentative indoor SPL of Item 7 is as follows:

49 53 47 36 28 16 -4 -24 dB

This can be compared with the desired NC-20

D	ata Shee	19					(Paragi	aph 6.9 Exampl
			Esti In	imated Indo cluding All	oor SPL at Ne Noise Reduc	lghboring Loc tion Contribut	ation, ions	
· 1	Vaise Sour	ce Conside	red in This	Data Sheet:	Exhaust of	200-hp Rec	ip. Engine	
	[			<u> </u>	Frequency Bar	nd in Hz		
	63	125	250	500	1,000	2,000	4,000	8,000
1	. Outdoor	SPL at Ne	ighboring L	ocation for N	io Noise Reduc	tion Contributio	ns (from 1tem 4 o	f Data Sheet 8):
	59	65	61	52	47	38	20	2
2.	Noise Rec	l fuction fr	om Outdoo	rs to Indoors	Provided by Ne	ighbor Building,	from Table 31:	L
					12		1	16
		10		1 12	13	14	<u> </u>	10
З,	Noise Hec	wetten Pr	ovided by 2	iound Barrier	, from Table 32	: 		
	0	٥	D	0	0	0	0	D
4.	Noise Red	luction Pr	ovided by V	Voods, from 1	Fable 33:			
	1	2	3	4	6	8	9	10
5.	Tentativo	Indoor SP	L to This P	oint (Item 5	= item 1 – item	n 2 - Item 3 - II	tem 4):	
	49	53	47	36	28	16	-4	-24
6.	Directivity	Effect of	Stack One	ninn. If Annii	icable, from Tat	nie 34:		
		1			11			
	0	0	0	0	0	0	0	
7.	Caution: C	Indoor SP Ibserve alg	L to This Po pebraic signs	int (ltem 7 : in combinin	= item 5 + item g items!	6}		
	49	53	47	36	28	16	-4	-24
8. /	Approximi irom Mutti	nte Noise i Ier Data S	Reduction of upplied by (	if Muffler Pla Manufacturei	nned for Use wi	ith Noise Source,	, if Applicable, fro	om Tables 35-37
. 1	10	15	13	11	10	9	8	8

. ۳. . . . . . . . . . . . . .

....

criterion for the residence, from Tables 1 and 2:

NC-20: 50 40 33 26 22 19 17 16 dB

This comparison shows that the tentative indoor SPL exceeds the NC-20 criterion by the following amounts:

-13 14 10 6 - - -dB

These excesses could be met by the use of a reactive muffler (Table 35) for the engine exhaust. A small-sized, low pressure-drop muffler, having the following approximate noise reduction

would be considered adequate to meet the requirement even though it is shy by  $1^{\circ}$  dB in the 250 Hz band, provided this is the only noise source to be considered. If there were several noise sources present, their total effect on the residence should be considered. And it might be necessary to use a larger muffler with greater noise reduction.

ł

For a gas turbine engine, the amount by which the tentative indoor SPL of Item 7 of Data Sheet 9 exceeds the desired noise criterion for that indoor area would give the noise reduction requirement for a muffler for either the air intake or gas exhaust of the engine. This requirement would then be specified to the engine installer or muffler supplier. A very rough idea of the muffler size might be obtained by checking the requirement with the muffler examples given in Tables 36 and 37.

In using Data Sheet 9, it is cautioned that if the noise reduction provided by a muffler was included in estimating the PWL of the noise source (as in Data Sheet 2, 3 or 4), then the noise reduction of that same muffler cannot be used again in Item 8 of Data Sheet 9.

If a very unusual amount of noise reduction is required or if it appears that conventional muffling devices cannot meet the requirement, re-check all calculations very carefully and, if necessary, obtain the assistance of the muffler company engineer or of an acoustical consultant. 6.10 Noise Control for an Outdoor Engine

If it is expected that an engine is to be located out-of-doors with no engine room or acoustic enclosure, the SPL of the engine casing noise at the desired distance should be determined with the use of Data Sheet 8 and appropriate tables of data. Then, with the use of Data Sheet 9, the indoor SPL values at the nearest neighbor of interest can be determined. These SPL values should then be compared with the appropriate indoor noise criterion levels for that neighbor. For a nearby residential neighbor in a quiet community situation, the engine noise might be found to be 10-30 dB too high for neighbor acceptance. Such a situation would clearly require that the engine be housed inside an engine room. As a rule-ofthumb, the TL of the walls of such an engine room should be at least 15 dB greater in all frequency bands than the noise excess found for the outdoor engine, and the TL of the roof of the engine room should be at least 6 dB greater in all frequency bands than the noise excess found for the outdoor engine.

These required TL values would apply if there is no acoustic absorption material inside the engine room. These wall and roof TLs can be reduced 3, 6, or 9 dB for acoustic absorption inside the engine room that meets condition 2, 3 or 4, respectively, in the footnotes of Table 23. Doors and windows (if the latter are necessary) should be compatible with the walls in terms of noise reduction (see Paragraph 5.7).

If an engine is expected to be enclosed in some type of lightweight weather enclosure, the same type of calculation as outlined above should first be carried out. If the lightweight enclosure does not meet the TL requirement, the construction should be improved so that it will meet the TL requirement.

### 6.11 Noise Escape from an Opening

Although one may try to minimize the escape of noise from a room, there are times when the noise will escape, and, at those times, it may be necessary to know the amount of escaping noise. An example of such a situation is the ventilation ducts that serve an engine room. In order to allow air movement through the room, the ducts may be open to the outside, and noise would escape through the open ducts unless suitable muffling is provided.

The power level of sound that passes through an opening into or out of a room is approximately

 $PWL (in dB re 10^{-12} watts) = SPL + 10 \log A - 10$ 

where SPL is the sound pressure level in the room at or near the opening and A is the cross-section area in square feet of the opening. A new term "Area Factor" ("AF") is defined as follows:

$$AF = 10 \log A - 10$$
.

Then,

ġ

$$PWL = SPL + AF$$

Table 38 gives a range of values of AF for a representative group of areas.

If an engine room has a large opening to the outside to provide ventilation air to the engine, the PWL of the noise escape through that opening should be determined. The SPL is taken as the engine room SPL, and the AF term is found from Table 38 for the area of the opening. The resulting PWL should then

STATISTICS STATES

be treated as a noise source that radiates to a neighbor, using Data Sheets 8 and 9 as applicable.

There is an important caution to be introduced at this point. In a small room with little or no acoustic absorption, the reverberant SPLs are quite high. For a large area duct or opening to the outside, it is possible to calculate a PWL of escaping noise that is actually larger than the PWL of the noise source inside the room in the first place. Of course, this is a fallacy of the calculations and cannot really exist. Therefore, when calculating the PWL of escaping noise through an opening, check back to the original PWL of the engine or noise source in the room to be certain that the PWL of the escaping noise is never greater than the PWL of the basic noise source inside the room.

Excess noise escaping through holes, ducts, open windows or ventilation ports can best be controlled with the use of a muffler in the opening. Table 39 gives the approximate noise reduction of typical "packaged duct mufflers" supplied by several acoustic products manufacturers. These mufflers may also be used in air-intake openings for turbine engines, if the air speed does not exceed the upper limits suggested by the muffler manufacturer. In fact, so that air flow noise itself is not excessive, the air speed through the muf-



Vertical exhaust silencer is used on Solar gas turbine engine to reduce noise radiated from this natural gas pumping station to the neighbors. (Photograph is courtesy of Solar Division of International Harvester Company.) fier passages probably should not exceed about 4,000 fi/min.

Noise escape through ducted openings can also be controlled by duct lining. This is not discussed in detail here, since duct treatments can be referred to in the "ASHRAE Guide and Data Book" (see Footnote 2 under Paragraph 4.2 of the manual).

### 6.12 Caution on Outdoor Engine Noise

In Paragraph 6.3, it was stated: "If the noise, when heard indoors in the neighbor building, can be made to be no greater than the criterion levels of Table 2 for the Table 1 activity areas, it is quite likely that there will be no complaint against the noise," This is generally true, but it is also important to realize that neighbors who value their outdoor surroundings may not tolerate very much outdoor audible noise even if it is essentially inaudible when heard indoors. For such situations it would be well to compare the outdoor noise made by the engine with the ambient or background noise for the area. If actual measured noise levels cannot be made, an approximate estimate of outdoor nighttime background noise for any area can be determined from a table and set of curves given in the latest "ASHRAE Guide and Data Book" in the chapter on noise control. In general, critical neighbors would not willingly accept outdoor

J

engine noise levels that are more than about 5 to 10 dB above the nighttime ambient noise levels. If the outdoor engine noise is above this amount, additional noise control should be included until the engine noise levels equal or are only slightly above outdoor ambients at the neighboring location.

### 6.13 Noise Codes and Ordinances

A major objective of this manual is to provide the guidance necessary to achieve a sufficiently quiet engine installation so that neighbors will not complain of the noise. If this objective is met, there may be no reason to invoke a noise ordinance against such an installation. Nevertheless, where local codes or ordinances exist, it is desirable to check the expected noise levels of the engine installation, including all the noise control designs, to determine if they conform to the code requirements. If additional noise reduction is shown to be necessary, determine the additional treatments that must be incorporated into the total design. Data Sheet 8 may be used to determine the outdoor noise levels at any specified distance from the total noise source, and, where applicable, the noise reduction steps of Items 3, 4, 6 and 8 of Data Sheet 9 and other treatments can be used to estimate the ultimate noise levels to be expected at the point specified by the ordinance.



Arrangement of mulflers control engine exhaust noise for bank of engine-driven compressors at this gas transmission line compressor station. (Photograph is courtesy of Universal Silencer Corporation.)



のない見れていた。それのためのないのであったのです。「ないない」ではないで、

日本町の日本の人でいます

ų

When numerical values are assigned to noise criteria, PWLs, SPLs, TLs, NRs, etc., this raises the question of accuracy of data and possible tolerances on the solutions. Each estimate may involve an accuracy of 2 or 3 dB, and the cumulative effect of several estimates could produce an over-design or an underdesign of a few decibels. The results, however, have been based on many measurements and many experiences with engine noise sources and noise control problems. It is believed that the methods and procedures given in the manual will produce reliable and workable acoustic solutions to most engine installations.

In general, the noise estimation procedure used in this manual will tend to over-estimate the noise of an "average" engine by about 2 or 3 dB, in order to protect the designs against many of the engines that are just slightly noisier than "average." It is believed that this degree of over-estimation will yield noise control designs that will encompass about 85% to 95% of all diesel and gas reciprocating engines and about 75% to 90% of all gas turbine engines. The remaining reciprocating engines may fall as high as 2 or 3 dB above this "design group" (possibly higher for turbocharger air-intake noise, which is usually not a major problem, however), and a few of the remaining gas turbine engines may fall as high as 5 to 10 dB above this design group.

Acoustic designs are rated here in four classes as to their relative ability to meet calculated needs. These four classes are described as follows:

- 1. A design may be rated *preferred* if it equals or surpasses the noise requirements of the analysis in all frequency bands.
- 2. A design may be considered *acceptable* if it produces no more than the following noise excesses above the design goal:

4 dB in the 63 and 125 Hz band, 3 dB in the 250 Hz band, or 2 dB in all higher frequency bands.

3. A design should be considered *marginal* if . it produces the following noise excesses above the design goal:

5-7 dB in the 63 and 125 Hz band,

4-6 dB in the 250 Hz band, or 3-5 dB in all higher frequency bands.

4. A design should be considered *unacceptable* if it produces any higher noise excesses than those listed immediately above in the "marginal" category.

The above ratings and tolerances take into account the general accuracy of data fed into the analysis as well as the "average" reaction of people to noise.

A basic philosophy on two points is offered. First, if a design involves a permanent structure, such as a building wall or floor, that is not easily modified after construction, use a conservative approach and avoid relaxing the design decisions. On the other hand, if a design involves a portable or replaceable item, such as a muffler or a movable partition, that can be modified or corrected later if necessary at relatively small extra cost, then, if desired, design compromises might be made. Second, if the client, customer or building owner is willing to support a step-by-step noise reduction program in an attempt to "get-by" with the least expensive design, it might be justifiable to try some compromise decisions in pursuit of an acceptable solution. (It is cautioned that this may not result in the least expensive total solution, however.) On the other hand, if the client does not wish to be bothered with attempts at intermediate solutions, then the conservative approach should be followed.

If a compromise solution is carried out, there is the possibility that additional noise control may be needed later; however, if no complaint arises from the compromise decision, possibly a less expensive solution will have been reached.

It is pointed out here, in addition to being mentioned in Paragraph 5.3, that when several noise sources are present, their total noise must be considered in any complete evaluation. If just three noise sources, say, the casing, inlet and exhaust of a turbine engine, were each controlled to yield alone an NC-25 at a critical neighboring location, the total noise of those three sources might equal an NC-30 condition. Thus, when multiple sources or paths of noise are present, allow adequate excess noise control to assure that the final total still meets the desired criterion goal.

### Section 8

VIBRATION ISOLATION OF RECIPROCATING ENGINES

In this section, vibration isolation of reciprocating engines is discussed for two general locations: (1) on an on-grade slab, such as in a basement or ground floor location, and (2) on an upper-floor of a multi-floor building. All suggestions given here are based on acoustical considerations only; these are not intended to represent structural design requirements. These suggestions apply to both the engine and all attached equipment driven by the engine and apply equally to rotary- or reciprocating-action driven equipment. It is assumed that the mechanical engineer, structural engineer, or equipment manufacturer will specify a stiff, integral base assembly for the mounting of the equipment and that all equipment will be properly aligned. The base assembly should be stiff enough to permit mounting of the entire equipment load on individual point supports, such as "soft" steel springs.

### 8.1 On-Grade Location

There is no limit on the size of the recipro-

cating engine that can be installed at an ongrade location.

It is conventional practice, in the installation of large gas or diesel reciprocating engines, to install the engine assembly directly and securely onto a large concrete inertia block having a weight of about 2 to 5 times the total weight of the supported load and having sufficient stiffness to provide the desired degree of alignment. The engine manufacturer will usually specify the minimum dimensions for the concrete block and the maximum tolerances for alignment.

It is recommended here that an engine assembly on-grade be installed rigidly on a large and massive concrete inertia block having a weight not less than twice the weight of the total supported load; even greater weight is desirable. This mass adds stability to the machine and reduces the vibration of the entire assembly. The following schedule is offered as a guide for vibration isolation of the total "assembly," which is here taken to consist of the engine, the driven equipment, and the concrete inertia block (here called "engine base") to which these pieces are rigidly mounted. The term "engine assembly" denotes the engine and all the equipment that it drives. Distances named in this schedule may be either horizontal or vertical distances in the earth, in a building structure or in a combination of both.

The second states and the second states and the second states and the second states and the second states and the

- A. For engines at or under 600 rpm or over 1,200 hp:
  - No vibration isolation of the assembly is required if there is no Category

     (Table 1) area within 500 feet, or no Category 2 or 3 area within 250 feet, or no Category 4 or 5 area within 150 feet of the engine base. It is good practice, nevertheless, to give the engine base its own footings, separated from the footings of the engine room, with a structural break between the floor slab or floor grille of the engine room and the engine base.
  - 2. For distances closer than those listed immediately above, for the indicated Categories of Table 1, the engine base should be supported on steel spring vibration isolation mounts that have a static deflection<sup>1</sup> of at least 2 inches for engine speeds of 301 to 600 rpm or at least 3 inches for engine speeds of 200 to 300 rpm. For any isolated base, there must be no rigid structural connections between the engine assembly and the building proper.
  - 3. In addition, the steel springs listed above should rest on pads of compressed glass fiber, ribbed or wafflepattern neoprene, if the engine assembly is located within 200 feet of a Category 1 area, within 100 feet of a Category 2 or 3 area or within 50 feet of a Category 4 or 5 area. (It is

assumed throughout this schedule that feelable vibration is acceptable in Category 6 areas. If this is not an acceptable assumption, Category 6 should be considered along with Categories 4 and 5.)

- B. For engines above 600 rpm and under 1,200 hp:
  - No vibration isolation of the assembly is required if there is no Category 1 area within 300 feet, no Category 2 or 3 area within 150 feet, or no Category 4 or 5 area within 75 feet of the engine base. It is good practice, nevertheless, to give the engine base its own footings, separated from the footings of the engine room, with a structural break between the floor slab or floor grille of the engine room and the engine base.
  - 2. For distances closer than those listed immediately above, for the indicated Categories of Table 1, the engine base should be supported on steel spring vibration isolation mounts that have a static deflection of at least 2 inches for engine speeds of 601 to 1,200 rpm or at least 1 inch for engine speeds above 1,200 rpm. For any isolated base, there must be no rigid structural connections between the engine assembly and the building proper.
  - 3. In addition, the steel springs listed above should rest on pads of compressed glass fiber or ribbed or waffile-pattern neoprene, if the engine assembly is located within 200 feet of a Category 1 area, within 100 feet of a Category 2 or 3 area or within 50 feet of a Category 4 or 5 area. (It is assumed throughout this schedule that feelable vibration is acceptable in Category 6 areas. If this is not an acceptable assumption, Category 6 should be considered along with Categories 4 and 5.)
- C. For engines above 1,200 rpm and under 400 hp:
  - 1. All vibration isolation conditions are

<sup>&</sup>lt;sup>1</sup>"Static deflection" of an isolation mount is the distance that the mount will compress or deflect when the full weight of the load is applied. The mounts should be selected and located so that all mounts in a given system compress approximately uniformly all around the base and each mount meets the required minimum static deflection. The isolation mount manufacturer will usually recommend specific mounts with his submittal on a job.

the same as in Group B immediately above, except that the concrete inertia block can be eliminated, if desired. If the concrete block is eliminated, a substantial housekeeping pad should be provided under the engine assembly, and the engine assembly should be mounted on a rigid steel frame that is stiff enough to be supported off the floor on individual steel spring isolators without introducing stability or alignment prob-

lems. Even here, the concrete inertia block is recommended as a means of providing this stability and stiffness, but it is not absolutely necessary. Items 1, 2 and 3 of Group B above still apply throughout.

D. In this basic guide, wherever vibration isolation of the assembly is required, the following suggestions also apply:

1.



Gas reciprocating engine-generator set provides power for research facility. Engine assembly is thoroughly vibration-isolated from on-grade slab, and all piping and conduit have flexible connections. Note control room viewing window in rear wall. (Photograph is courtesy of Northern Illinois Gas Co.)

A ALTERNATION OF PARTICULAR

47

The second s

should be made up of three layers of the conventional pad-type material, giving a total thickness of approximately 1 inch of neoprene, with adjacent pads separated by sheet steel plates of approximately 1/16 inch to 1/8 inch thickness. Compressed glass fiber pads should have a thickness of at least 1 inch. The area of the pads should be such as to provide the surface loading recommended by the pad manufacturer. For critical locations, provision should be made to permit replacement of the pads after about 25 years, as the pad material may deteriorate by that time.

- 2. For an isolated engine assembly there should be no structural, rigid connections between the engine assembly and the building proper. This includes piping, conduit, and ducts to and from the assembly.
  - a. A long bellows-type thermal expansion joint in the exhaust piping meets this requirement, as does a flexible connection in the inlet-air ducting to the engine.
  - Piping to the engine assembly may contain long flexible connections (length at least 6 times the outside diameter of the piping) that are not short-circuited by steel bars that bridge the flanges of the flexible connections; or piping may be used without flexible connections, if the piping is supported on vibration isolation hangers or mounts for a distance along the pipe of at least 100 pipe diameters. The vibration isolation hangers should have a static deflection of at least one-half the static deflection of the mounts that support the engine base. If steel springs are used in the pipe hangers, neoprene or compressed glass fiber pads should be in series with the springs.
  - c. Electrical bus bars from the generator should either contain a 6-foot length of braided, flexible

conductor across the vibration isolation joint, or the rigid bars should be supported from resilient hangers for a distance of about 25 feet from the isolated assembly.

3. Where steel springs are used, unhoused stable steel springs are preferred. If housed or enclosed springs must be used, special attention must be given to the alignment of the mounts so that they do not tilt or bind in any direction inside their housings. Further, there should be some visual means to check the spring mount in its final position to be certain that binding or tilting does not take place.

### 8.2 Upper-Floor Location

It is strongly suggested that no reciprocating engine rated over about 1,000 hp or no reciprocating engine operating at a speed lower than about 900 rpm be installed in an upperfloor location. There are no known installations beyond these limits that give rise to these restrictions, but there is concern that the low-speed reciprocating action of very large engines might excite some of the resonances of various building structural elements and produce unwanted noise or vibration, even if the engine is vibration isolated. Then, too, a large concrete inertia block for such a large engine produces structural support problems for an upper-floor location. If engines above 1,000 hp or below 900 rpm must be located in an upper-floor location, it is advisable to engage the services of an acoustical consultant to assist with the design,

The following suggestions apply for any building containing Category 1-5 occupancy areas (see Table 1),

1. The entire engine assembly should be mounted rigidly to a concrete inertia block having a weight at least 3 times the total weight of the supported load. The concrete inertia block may be eliminated, if desired, for any engine of less than 100 hp that is located two or more floors away from a Category 1 or 2 area, or that is not located directly over a Category 3



and good alignment to the entire assembly. Its area should be at least as large as the overall area of the equipment that it supports. If the engine drives a refrigeration compressor which is connected directly to its evaporator and condenser cylinders, all this equipment should be mounted together onto the same concrete block. The bottom of the inertia block should rest at least 4 inches above the top of the structure slab. If a Type 5 floating-floor slab is involved, this 4-inch air space under the concrete inertia block should be covered with 2-inch-thick lowcost glass fiber or mineral wool as first mentioned under the Type 5 floor-ceiling combination.

If a concrete inertia block is not used, a substantial housekeeping pad should be provided under the engine assembly and the engine assembly should be mounted on a rigid steel frame that is stiff enough to be supported off the floor on individual steel spring isolators without introducing stability or alignment problems.

2. The concrete inertia block or the stiff steel frame of Item 1 above should be supported off the structure floor slab with steel spring vibration isolation mounts. The static deflection of the steel springs should be at least 3 inches for engines in the speed range of 900-1,200 rpm that are mounted on concrete inertia blocks and at least 2 inches for engines above 1,200 rpm mounted on concrete inertia blocks. Any engine mounted on a steel frame (in accordance with Item 1 above) should be supported off the floor on 2-inch static deflection steel springs.

3. Each steel spring should rest on a pad of



Waukesha gas engine drives refrigeration system equipment in this top-floor installation above office suites. Integral concrete inertia block is isolated from floor with steel springs in series with compressed glass fiber pads. (Photograph is courtesy of Consolidated Kinetics Corporation.) compressed glass fiber, or ribbed or wafflepattern neoprene.

- 4. To give a stiff support to a reciprocating engine, either with or without a concrete inertia block, the structural floor slab should be at least 10 inches thick and made up of dense concrete (140-150 lb/cu ft). In addition, the engine assembly should be located over primary or secondary beams supporting the floor slab, if at all possible.
- 5. All suggestions listed under Group D of Paragraph 8.1 apply here also.

If an engine assembly is mounted on the upper floor of a building containing only Category 6 occupancy areas, a concrete inertia block is not required for acoustical purposes but a stiff steel frame would be required for alignment purposes. The engine assembly should be vibration isolated on steel springs having at least 2-inch static deflection for 900-1,200 rpm engines or 1-inch static deflection for engine speeds above 1,200 rpm. Compressed glass fiber or neoprene pads are not required under the springs. The recommendations given under Item 2, Group D of Paragraph 8.1 would also apply here except that isolated pipe hangers can be limited to the confines of the engine room.

### 8.3 Special Situations

In the preceding paragraphs regarding ongrade locations and upper-floor locations, generalizations have been given that cover a wide range of situations. These generalizations will probably provide adequate protection for most typical equipment installations. Obviously, general rules cannot cover all marginal cases or complex situations, however. Thus, if unusual installations are planned or if some of the general suggestions or limitations given here cannot be met, it would be advisable to seek the assistance of an acoustical consultant who has worked on some of the specific installation problems. Vibration problems are sometimes quite complex and unpredictable.

# VIBRATION ISOLATION OF TURBINE ENGINES

Because of the high-speed rotary-action of the turbine engine, the engine alone does not induce very much vibration into its supporting floor slab. However, the equipment driven by the turbine engine may produce significant vibration. If a turbine engine is used to drive a piece of reciprocating equipment (such as a reciprocating compressor driven through a speed reduction gear), the entire assembly should be vibration isolated in accordance with the Section 8 suggestions for a reciprocating engine, where the speed and power considerations are based on the speed and power of the driven reciprocating device. For such a condition, the structure floor slab for an upper-floor installation should not be less than 8 inches of dense concrete (140-150 lb/cu ft).

For a turbine engine driving an entirely rotary-action load (centrifugal compressor or generator, with or without a gear), suggestions for vibration isolation are given in the paragraphs that follow, and the engine room floor slab should be not less than 6 inches thickness of dense concrete.

### 9.1 On-Grade Location

- 1. No vibration isolation of the assembly is required if there is no Category 1 area (see Table 1) within 200 feet, no Category 2 or 3 area within 100 feet, or no Category 4 or 5 area within 50 feet of the engine assembly. These distances should be measured along either horizontal or vertical directions in the earth, in building structures or in any combination of earth and/or structural paths.
- 2. If the engine assembly is located closer to

critical areas than the distances just given, the complete assembly should be supported off the floor slab on pads of 2-inch-thick compressed glass fiber or on pads made up of at least three layers of ribbed or waffle-pattern neoprene. Sheet metal spacers may be used, but are not required between adjacent layers of neoprene. Pipes, ducts and conduit to the assembly either should contain flexible connections or be supported from compressed glass fiber or neoprene-in-shear pipe hangers for a distance of at least 25 feet from the engine. The turbine engine manufacturer must approve the isolation mounting of the assembly because alignment is very critical for a turbine engine.

### 9.2 Upper-Floor Location

- 1. An engine assembly located on an upper floor (any floor above grade) of a multifloor building having Category 1 or 2 occupancy areas should be vibration isolated at least in accordance with Item 2 above, under Paragraph 9.1, including pipe, conduit and duct isolation.
- 2. If a Category 1 or 2 area is within about 30 feet, by structural path distances, of the engine room (either above, below or beside the engine room), stable steel springs having 1-inch static deflection should be inserted in the isolation mounting arrangement between the engine assembly and the glass fiber or neoprene pads on the floor.
- 3. If the building contains only Category 3-6 occupancy areas and a Category 3 or 4 area is within about 40 feet or a Category

5 area is within about 20 feet of the engine room, by combined horizontal and vertical structural path distances, the engine assembly should be vibration isolated in accordance with Item 2 of Paragraph 9.1, including pipe, conduit, and duct isolation.

4. If a Type 5 floor-ceiling combination (see Paragraph 5.8) is required for air-borne sound control to the floor beneath the engine room, a concrete inertia block is required under the engine assembly. The concrete should be at least 8 inches thick and it should be supported off the structure floor with 1-inch static deflection steel springs in series with 1-inch thickness of ribbed or waffle-pattern neoprene or compressed glass fiber pads. The engine assembly should be mounted rigidly onto this concrete inertia block and there should be at least 3-inch air space between the bottom of the inertia block and the top of the structure slab.

5. No vibration isolation of the turbine engine assembly is required if the building occupancy areas and distances from possible critical areas to the engine room are not covered by Items 1-4 immediately above.

For any of these upper-floor installations, proper airborne sound control should be provided in accordance with earlier parts of the manual, and the turbine engine manufacturer must approve the vibration isolation mounting of the installation.



Cas turbine and generator are mounted on steel springs and ribbed neoprene vibration-isolation mounts to reduce noise transmission to office area and executive conference room immediately below engine room, (Photograph is courtesy of Northern Illinois Gas Company.)



aunaser hender

This summary is intended primarily to serve as a reminder to not only be aware of, but seek out carefully each noise source and each noise path and provide for each an appropriate noise or vibration control treatment where necessary.

Each engine has three principal noise sources (the casing, the air intake and the gas exhaust). The casing noise, and sometimes the air-intake noise, is radiated directly into the engine room, and the engine room may adjoin a critical area of a building. The procedures of the manual will show the design options available for containing the noise with the use of suitable walls and floor-ceiling combinations, depending on the noise criterion requirements of the adjoining areas.

The engine exhaust noise, and sometimes the air intake noise and casing noise, is radiated out-of-doors and may be heard by the neighbors, depending on the distance away and the nature of the noise criteria applicable to the neighbors. Certain noise reduction effects may be determined, when they are applicable to the situation, and the need or the effectiveness of intake and exhaust mufflers can be calculated from the procedures given in the manual.

In some situations, different neighbors may have different noise criteria at different distances. It may be necessary to work out tentative solutions on paper for each critical neighbor and then to select the design solu-

tion that will encompass all neighbor situations. A thorough analysis must not ignore any of the noise sources, paths, or receivers.

12 33

Finally, vibration isolation is an important requirement for almost any engine assembly located near a critical occupancy area. Although vibration data are not generally susceptible to calculation and prediction, various vibration isolation suggestions are keyed into engine size, speed and location. Just as a bucket with many holes will leak water until all the holes are stopped up, so a noisy engine will leak noise until all its paths are identified and treated. This manual has attempted to help identify and quantify the noises so that they can be located and treated even before the building is built or the engine installed. Although some of these engines are admittedly noisy, their noise can be controlled quite easily when the right steps are taken.

			Stimated S Reciproc	ound Power L ating Engine (	evel (PWL) of Casing Noise	ſ		
1. Continue	ous Rating (	of Engine:_		hp or	k	w	•	
2. Engine S	ipeed;		_ rpm				,	
ı 3. Fuel: G	as Only 🗖	Liquid	i Only 🗖	Gas and Liqu	ild 🗆			
	1			Frequency Ban	d in Hz			7
63	125	250	500	1,000	2,000	4,000	8,000	
5 Sayard Co		om Table 3	for Itam 2 Sr	anad:				]
5. Speed Co	prrection fro	om Table 3 n Table 3 fo	for Itom 2 Sp pr Itom 3 Fuc	beed:				]
5. Speed Co	prrection fro	am Table 3 m Table 3 fo	far Itam 2 Sp pr Itam 3 Fuc	breed:				
5. Speed Co 6. Fuel Cor 7. Estimate Caution:	prrection fro rection fron d PWL (in o Observe alg	om Table 3 m Table 3 fo dB re 10–12 gebraic signs	far Itam 2 Sp pr Item 3 Fuc watt) of Cas i in combinin	sing Noise: (Iten	n 7 = Item 4 + 11	em 5 + Item 6).		
5. Speed Co 6. Fuel Cor 7. Estimate Caution:	prrection fro rection fron d PWL (in o Observe als	om Table 3 n Table 3 fo dB re 10-12 gebraic sign	far Itam 2 Sp pr Item 3 Fuc watt) of Cas i in combinin	sing Noise: (Iten g items!	n 7 = Item 4 + II	lem 5 + Item 6}.		
5. Speed Co 6. Fuel Cor 7. Estimate Caution:	rection from rection from d PWL (in c Observe als	om Table 3 n Table 3 fo dB re 10-12 gebraic sign	for Itom 2 Sp or Itom 3 Fuc watt) of Cas in combinin	beed:	n 7 = Item 4 + 11	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		
5. Speed Co 6. Fuel Cor 7. Estimate Caution:	rection from Prection from PWL (in c Observe als	om Table 3 n Table 3 fo dB re 10-12 gebraic sign	for Itom 2 Sp or Itom 3 Fuc watt) of Cas in combinin	sing Noise: (Iten g items!	n 7 = Item 4 + 11	iem 5 + Item 6}.	·	
5. Speed Co 6. Fuel Cor 7. Estimate Caution:	d PWL (in c Observe als	om Table 3 m Table 3 fo dB re 10-12 gebraic sign	for Itom 2 Sp or Itom 3 Fuc ? watt) of Cas in combinin	beed:	n 7 = Item 4 + 11	lem 5 + Item 6}.		
5. Speed Co 6. Fuel Cor 7. Estimate Caution:	correction from rection from PWL (in construction Observe als	om Table 3 m Table 3 fr dB re 10-12 gebraic sign	for Itom 2 Sp or Itom 3 Fuc ? watt) of Cas in combinin	beed:	n 7 = Item 4 + 11	lem 5 + Item 6).	· · · · · · · · · · · · · · · · · · ·	

S,

STER IN STREET

- )

			I	Estimated S	ound Power I	evel (PWL) of	[	
			Unm	uttied Keci	procating En	gne Exhaust r		
1. 0	ontinuo	us Rating	of Engine:		hp or		_kw	
2 /	ie testak	a. Turbor	haraar 🗖	No Turbo	charoar 🗍			
2. 7	AIF (IILAN)	e: 101000	nniða. ***					
3. E	ixhaust f	<sup>o</sup> ipe Lang	ih:	ft.				
Ľ					Frequency Band	in Hz		
	63	125	250	500	1,000	2,000	4,000	8,00
ц. 5. т	u bocha	rger Corre	rction from 1	able 4 for Ite	m 2 Air Intake			
ſ					T			
6. E	xhaust P	ipe Lengt	h Correction	from Table 4	for Item 3 Len	gth:		
Г	T					· · · · · · · · · · · · · · · · · · ·		
L					A	I		
	stimated aution: (	Diserve al	dB re 10772 gebraic signs	watt) of Unr in combining	nuffled Exhaust ) items!	Noise: (Item 7	= Item 4 + Item !	5 + Item 6)
7. E C			· · · · · · · · · · · · · · · · · · ·	1				. <u>.</u>
7. E C							1	

1	Data She	et 3						
-					~ 15	a statute s		
		-	А	of Untrea	sound Powe ted Turbocha ming of Reci	r Level (PWL) arger Noise at procating Engi	ine	
				<b>-</b> -				
1	. Continu	ous Rating	of Engine:_		hp or	<u>.                                    </u>	KW	
2	. Inlet Ai	ir Duct Len	ngth:	ft,				
					Frequency Ban	d in Hz		
	63	125	250	500	1,000	2,000	4,000	8,000
7	Dura Di	u fram Tr	bia E fas isa	m 1 Ostinas				
-				an i Astroj:	•.			
	L	I	l	<u> </u>		L		<u></u>
4	. Infot Ai	r Duct Len	igth Correcti	on from Table	e 5 for item 2 L	.ongin:		
	1	1						
	[	<u> </u>						
5	. Estimati Caution	ed PWL (in : Observe a	dB re 10-1: Ngebraic sign	<sup>2</sup> watt) of Units in combining	reated Turboci g items!	harger Noise: (It	em 5 =  tem 3 +  t	em 4)
5	. Estimate Caution	ed PWL (in : Observe a	dB re 10-1: Ngebraic sign	<sup>2</sup> watt) of Un in combinin	treated Turboci g itemsi	harger Noise: (It	em 5 = 1 tem 3 + 1 t	em 4)
5	Estimate Caution	ed PWL (in : Observe a	dB re 10-1: Igobraic sign	<sup>2</sup> watt) of Un ns in combinin	treated Turbocl g items!	harger Noise: (It	em 5 = item 3 + it	em 4)
ŝ	Estimate Caution	ed PWL (in : Observe a	dB re 10-12 Igebraic sign	<sup>2</sup> watt) of Un is in combinin	reated Turbocl g itemsl	harger Noise: (It	em 5 = Item 3 + It	em 4)
5	Estimat Caution	ed PWL (in : Observe a	dB re 10-11 ngebraic sign	<sup>2</sup> watt) of Un ns in combinin	reated Turboci g itemsi	harger Noise: {It	em 5 = 1tem 3 + 1t	em 4)
5	Estimat Caution	ed PWL (In : Observe a	dB re 10 <sup>-12</sup> Igobraic sign	2 watt) of Un ns in combinin	reated Turboci g itensi	harger Noise: (It	em 5 = )tem 3 + )t	em 4)
5	Estimat Caution	ed PWL (In : Observe a	dB re 10 <sup>-13</sup> Igobraic sign	2 watt) of Un ns in combinin	treated Turbocl g items!	harger Noise: (It	em 5 = )tem 3 + )t	em 4)
5	Estimate Caution	ed PWL (in : Observe a	dB re 10-12 Igobraic sign	<sup>2</sup> watt) of Un is in combinin	treated Turbocl g items!	harger Noise: (It	em 5 = )tem 3 + )t	em 4)
5	Estimate Caution	ed PWL (in : Observe a	dB re 10-12 Igebraic sign	<sup>2</sup> watt) of Units in combinin	treated Turbocl g itemsl	harger Noise: (It	em 5 = item 3 + it	em 4)
5	Estimation	ed PWL (in : Observe a	dB re 10-12 Igebraic sign	<sup>2</sup> watt) of Un is in combinin	ireated Turbocl g itemsi	harger Noise: (It	em 5 = item 3 + it	em 4)
5	Estimat Caution	ed PWL (in : Observe a	dB re 10-12 Igebraic sign	2 watt) of Units in combinin	treated Turbock g itemsi	harger Noise: (It	em 5 = Item 3 + It	em 4)

.

	Data :	Sheet 4						Shect 1 of
			Esti Exhi	mated Sound	d Power Leve ike Noise of (	el (PWL) of Ca Gas Turbine E	sing. ngine	
•	1, Cont	tinuous Rating	of Engine:_	·	hp or	k	w	
	2. Engi	ne Casing Cove	r: None 🗆	or Type		from Table 8		
	3. Exhi	oust Duct: R	iound 🛛, i	Rectangular D	], Longth	f		
	Duct	Lining: T	Ype	fro	m Table 9			
	Duct	: Turns: N	10,	Ty	pe	from Tabl	• 10	
	4. Intal	ce Duct: R	ound 🛛, I	Røctangular E	], Length	f	l.	
	Duct	: Lining: T	уре	fro	m Table 9			
	Duct	Turns: N	0	Tyl	pe	from Tabl	e 10	
	6. Exh	iust Muffler Su	pplied: Ye	15 🗆 No				
	6. Intai	ke Muffler Supj	plied: Ye	s 🖾 🛛 No	۵			
	6	3 125	250	500	Frequency Ban 1,000	d in Hz 2,000	4,000	8,000
						<b></b>		
	7. PWI	(in dR to 10−1	2 watt) of C	asina Noise fr	om Tabla 6 foi	r Item 1 Bating:		
	// / WE							·
	L			l		L		
	8. Noise	Reduction Pro	ovided by C	asing Cover (i)	I Any) of Item	2 as Estimated i	n Table 6 Footnoi	tø:
	L							 
		of Casing Noise	e with Cove Jebraic sign:	r (If Any): Ite in combining	ım 9 = Item 7 ·   items	+ Item B	-	
	9. PWL Cauti					r	·	
	9. PWL Cauti							
	9. PWL Cauti					}		(Contin

-----

......

<u>}\_\_</u>

al di Nga

Start Street

11.11

- j. •

• .

	la Sheet	4 (conti	nued)					Sheet 2 of 2
					Frequency Ban	d in Hz		
	63	125	250	500	1,000	2,000	4,000	8,000
•		- -			<u></u>		•	· · · · · · · · · · · · · · · · · · ·
0. 1	WL (ind	IB re 10~1	<sup>2</sup> watt) of E	xhaust Noise	from Table 7 f	or item 1 Ratin	0:	
ſ	·1		I			<u> </u>		
I			I					[
1. /	Attenuati Vestuda t	ion Provide	ed by Exhau	st Duct and T	furns of Item 3,	from Tables 9 a	and 10 or "ASHR.	AE Guide''
ſ	include r		ature correc		··			
l								
2. (	disertion	Loss of Ex	chaust Muffl	er of Item 5.	if Provided (us	e muffier manuf	iacturer's data for	appropriate
1	ands, co	rrected for	r exhaust ter	nperature):		r		
		_						
L				I	I			<u> </u>
3. F (	WL of E Caution:	xhaust No Observe al	ise Out of D gebraic signs	uct and Muff	ler (as Applicat 1 items!	ole): Item 13 = 1	tem 10 + Item 11	+ Item 12
ſ		1						·····
[	[							····
4. F	WL (in d	iB re 10 <sup>-1</sup>	<sup>12</sup> watt) of A	Nir-Intake No	ise from Table 8	3 for Item 1 Rat	ing:	
4. F	₩L (in d	16 re 10 <sup>-1</sup>	<sup>12</sup> watt) of A	Nir-Intoke No	ise from Table 8	3 for item 1 Rat	ing:	·····
4. F	WL (in d	iB re 1D-1	<sup>12</sup> watt) of A	Air-Intake No	ise from Table 8	3 for item 1 Rat	ing:	······
4. F	PWL (in d	iB re 1D-1	<sup>12</sup> watt) of A	Nir-Intake Noi	ise from Table 8	3 for item 1 Rat	ing: d 10 or "ASHRAE	E Guide":
4. F 5. A	PWL (in d	iB re 10-1 on Provide	<sup>12</sup> watt) of A	Nir-Intoke Noi	ise from Table 8	3 for Item 1 Rat	ing: d 10 or "ASHRAE	E Guide":
4. F [ 5. /	PWL (in d	IB re 10 <sup>-1</sup>	22 watt) of A	Nir-Intake Noi	rns of Item 4, fi	3 for Item 1 Rat	ing: d 10 or "ASHRAE	E Guide":
4. F 5. A 6. L	PWL (in c	iB re 10 <sup>-1</sup>	12 watt) of A ed by Intake	Nir-Intake Noi Duct and Tu of item 6, If	rns of Item 4, fo Provided (use r	3 for item 1 Rat	ing: d 10 or "ASHRAE turer's data for ap	E Guide":
4. F 5. A 6. I	PWL (in c	iB re 10-1 on Provide Loss of Int Table 39	22 watt) of A ed by Intake take Muffler for example	Air-Intake Noi Duct and Tu of Item 6, If ):	rns of Item 4, fo	3 for Item 1 Rat rom Tables 9 an	ing: d 10 or "ASHRAE turer's data for ap	E Guide": opropriate
4. F [ 5. A 6. [	PWL (in c	iB re 10 <sup>-1</sup> on Provide Loss of Int Table 39	12 watt) of A	Nir-Intake Noi Duct and Tu of item 6, If	ise from Table 8	3 for item 1 Rat rom Tables 9 an nufflør manufac	ing: d 10 or "ASHRAE turer's data for ap	E Guide":
4. F [ 5. /	PWL (in d	iB re 10 <sup>-1</sup> on Provide Loss of Int Table 39	2 watt) of A ed by Intake take Muffler for example	Nir-Intake Noi Duct and Tu of item 6, If ):	rns of Item 4, fr Provided (use r	3 for item 1 Rat rom Tables 9 an nufflør manufac	ing: d 10 or "ASHRAE turer's data for ap	E Guide":
4. F [ 5. / 6. [ 7. P	WL (in c	(B re 10 <sup>-1</sup> on Provide Loss of Int Table 39 stake Noise Item 14 +	2 watt) of A ed by Intake take Muffler for example	Air-Intake Noi Duct and Tu of Item 6, If ): t and Muffler tem 18 in combining	rns of Item 4, fi Provided (use r	3 for Item 1 Rat	ing: d 10 or "ASHRAE turer's data for ap	E Guide":
4. F [ 5. /	WL (in c	IB re 10-1 on Provide Loss of Int Table 39 stake Noise Item 14 + Observe alg	2 watt) of A ed by Intake take Muffler for example b Out of Duc I tem 15 + In gebraic signs	Vir-Intake Noi Duct and Tu of item 6, If ): tand Muffler tem 16 in combining	ise from Table 8 rns of Item 4, fi Provided (use r (as Applicable ) items!	3 for item 1 Rat	ing: d 10 or "ASHRAE turer's data for ap	E Guide":

1

AND A CANADA

「日本のない」というという

いたいからい、ならいないのかりましたとう

\*\*\*\*\*\*

÷

 $\mathbf{r}_{i}^{(t)}$ 

anagaran an

Data S.	nect	Э
---------	------	---

Estimated Sound Pressure	Level (SPL	L) in an Engine Room	
--------------------------	------------	----------------------	--

The second states

1. Volume of Engine Room:\_\_\_\_\_ft. X \_\_\_\_\_ft. X \_\_\_\_\_ft. = \_\_\_\_\_cu, ft.

2. Total Interior Surface Area of Room (including floor) =\_\_\_\_\_\_sq, ft,

3. Area of Acoustic Treatment \_\_\_\_\_\_sq. ft.

4. Per cent Area Covered by Acoustic Treatment (Item 3/Item 2) X 100 =\_\_\_\_%

5. Thickness of Absorption Material: 3/4 in.-1 in. 🖾 1-1/2 in.-2 in. 🗔

6. Acoustic Treatment Condition of Room from Items 4 and 5, as Defined in Footnotes of Table 11: Condition 1 Condition 2 Condition 3 C Condition 4 C

7. Room Correction Term from Table 11 for Items 1 and 6:\_\_\_\_\_dB

Frequency Band in Hz									
63	125	250	500	1,000	2,000	4,000	8,000		

8. Total PWL of All Noise Sources in Engine Room (from Item 7 of Data Sheet 1 for Reciprocating Engines or from Item 9 of Data Sheet 4 for Turbine Engines):

 •			
		7	
	•		

9. Reverberant SPL in Engine Room (Item 9 = Item 8 + Item 7; Item 7 Value Is Same for All Frequency Bands and Is Zero or a Negative Quantity for All Room Conditions) Caution: Observe algebraic signs in combining terms!

	•			

10. Approximate SPL Near Engine (Item 10 = Item 9 + 3 dB)


### Data Sheet 6

### Sound Transmission to Adjoining Room Through Common Wall

Sound Transmitting Room

Sound Receiving Room

TT ALTOTAL HALL TT TO DAN FOR

1. Area of Common Wall That Transmits Noise: \_\_\_\_\_\_sq. ft.

2, Total Interior Surface Area of Receiving Room (including floor): \_\_\_\_\_\_sq. ft.

3. Total Interior Surface Area Divided by Area of Common Wall: \_\_\_\_\_\_ (Item 3 = Item 2 ÷ Item 1)

4, Area of Acoustic Treatment in Receiving Room\*: \_\_\_\_\_\_ sq. ft.

5. Per cent Area of Receiving Room Covered by Acoustic Treatment \_\_\_\_\_% (Item 5 = 100 X (tem 4 ÷ Item 2)

6. Thickness of Absorption Material: 3/4 in.-1 in. 🖾 1-1/2 in.-2 in. 🖾

7. Acoustic Treatment Condition of Receiving Room from Items 5 and 6, as Defined in Footnotes of Table 23: Condition 1 [], Condition 2 [], Condition 3 [], Condition 4 []

8. Wall Correction Term "C" from Table 23 for Item 3 and Item 7 Conditions: \_\_\_\_\_\_dB

	Frequency Band in Hz									
63	125	250	500	1,000	2,000	4,000	8,000			

9. Expected Wall Construction Material:\_

1 14 117

10, Estimated "TL" of Wall from Tables 13-22 or Other Source

		1						
stimat	ed SPL in	 Sound Tran		 m from Data \$		m 9 Reverbera	nt SPL If Source Is	5
. or M	ore from C	ommon Wa	ll or Item 10	Close-in SPL I	f Source Is Les	Than 5 ft, fro	om Common Wall).	
		Raceivina R		3.≃itam. 12⊫			k	
stimat			DOW LIGHT .					
stimat		-r	<u> </u>					

	Sheet	7	S	ound Trans	smission to A	dioining Roon	n	
			_	Through	Common Flo	or-Ceiling	-	
Sound	i Transr	nitting R	om				Sound I	Receiving Ro
1. A	res of C	ommon f	 loor-Ceiling	That Transm	its Noise:	K		
2. Te	otal inti	rior Surf	ace Area of	Receiving Ro	om (including i	lioor):	sq. ft.	
3, Ta (ii	otal Inte tern 3 =	rior Surf. Item 2 ÷	ce Area Div Item 1)	ided by Area	of Common Fi	oor-Cailing:		
4. A	røa of A	coustic T	reatment in	Receiving Ro	om":		t.	
5. Pc (I1	r cent / tem 5 =	Area of Ri 100 X Iti	eceiving Roc em 4 ÷ Item	m Covered b 2)	y Acoustic Tra	stment		
6. TI	nicknes	of Absor	ption Mater	ial: 3/4 (n	-1 in. 🗆 1	•1/2 in.—2 in. 🗇		
7. Ac Cc	coustic ' andition	Treatmon 1 🗖 ,	t Condition Conditio	 of Receiving n 2 □,	Room from Ite Condition 3 🗆	ms 5 and 6, as D , Conditio	afined in Footnat	tes of Table 2
8. Fi	oor Car	rection T	erm "C" fro	m Tøble 23 fi	or Item 3 and 1	tem 7 Condition	5;	dB
F					Frequency Ban	id in Hz	 	 
	63	125	250	500	1,000	2,000	4,000	8,000
	proxim terpola	ate "TL" te Betwee 25-28):	of Floor-Co In TL Value	tiling Type of s Shown, If C	l Item 9 for Neg Desired, But Do	Not Exceed TL	Dimensions, from Values Shown in	Tables 25-29 Right-Hand C
10. Aj (li of	Tables .			(	•		1	<u>l</u>
10. Aj	Tables			<u> </u>				
10. Aj (); of 11. Ei ();	Tables itimated tem 11 -	"NR" of = Item 10 Observe a	Floor-Ceili + Item 8): Igebraic sign	ng in Equatio Is in combini	i on NA = TL + C, og items!	, Using Values o	f Items 8 and 10	
10. Aj of 11. Ei 11. Ei Ci	Tables itimated tem 11 -	- "NR" of I tem 10 Observe a	Floor-Celli + Item 8): Igebraic sign	ng in Equations in combining	 	, Using Values o	f Items 8 and 10	}
10. Aj (ii of 11. Ei (ii) Ci 12. Ei A	Tables	I "NR" of I tem 10 Observe a I SPL in S Engine (	Floor-Celli + Item 8): Igebraic sign ound Trans or Item 10 (	ng in Equations is in combinin is in combinin is in combining mitting Roon Close-in SPL f	n NR = TL + C ng items! n from Data Shi or Floor Under	, Using Volues o eet 5 (Uso Item s Engino):	f Items 8 and 10 f Bevorberant SPL	for Floor-C
10. Aj (ii of 11. Ei (ii Ci 12. Ei A	Tables	"NR" of = Item 10 Observe a I SPL in 5 > Engine d	Floor-Celli + Item 8): Igebraic sign ound Trans or Item 10 (	ng in Equations in combinit Is in combinit mitting Roon Close-in SPL f	n NR = TL + C ng items! from Data Shi or Floor Under	, Using Volues o eet 5 (Use Item 1 Engine):	f Items 8 and 10 B Revorberant SPL	for Floor-C
10. A) (I) of 11. E: (I) (I) (I) (I) (I) (I) (I) (I) (I) (I)	Tables itimated itim 11 itimated itimated bove the itimated	- "NR" of = Item 10 Dbserve a P SPL in S = Engine ( SPL in F	Floor-Celli + Item 8): Igebraic sign ound Trans or Item 10 ( leceiving Rd	ng in Equations in combining from the second	n NR = TL + C ng items! n from Data Shi or Floor Under	, Using Volues o eet 5 (Use Item 1 Engine): em 11)	f Items 8 and 10 9 Revorberant SPL	. for Floor-Co
10. Aj (li of 11. Ei (li Cu 12. Ei A 13. Ei	Tables itimated itimated itimated itimated itimated itimated	"NR" of = Item 10 Observe a SPL in 5 = Engine ( SPL in F	Floor-Celli + Item 8): Igebraic sign ound Trans or Item 10 ( Ieceiving Re	ng in Equations in combinin mitting Roon Close-in SPL f	ng itemst n from Data Shi or Floor Under	, Using Volues o eet 5 (Uso Item ( Engino): em 11)	f Items 8 and 10	- for Floor-C

.

ł

4

### Data Sheet 8

### Estimated Outdoor Sound Pressure Level (SPL) Due to an Outdoor Sound Source PWL

1. Distance to Noise Source:\_\_\_\_\_\_ft.

Frequency Band in Hz								
63	125	250	500	1,000	2,000	4,000	8,000	

### 2. Total PWL of All Outdoor Noise Sources at Source Position:

				•	
		4			
				r	
	•				
		•			

3. Outdoor Distance Term from Table 30 for item 1 Distance:

		•	
	1 1		

4. Approximate Outdoor SPL at Distance of Item 1: (Item 4 = Item 2 -- Item 3) Caution: Observe algebraic signs in combining items!

A			 	

### Data Sheet 9

## Estimated Indoor SPL at Neighboring Location,

				Frequency Ban	d in Hz		<u> </u>
63	125	250	500	1,000	2,000	4,000	8,000
)utdoor	SPL at No	ighboring L	.ocation for l	No Noise Reduct	ion Contributio	ons (from Item 4	of Data Sheet 8):
			·	1		1	
loise Ro	duction fro	om Outdoo	is to Indoors	Provided by Ne	ighbor Building	, from Table 31:	
			<u> </u>	1	· · · · · · · · · · · · · · · · · · ·	1	
loise Re	duction Pr	ovided by S	Sound Barrie	r, from Table 32		· · · · · · · · · · ·	_,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
			}			T	1
ioise fio	duction Pr	ovided by \	Noods, from	Table 33:			
			<u> </u>	1	<u>,                                     </u>	1	
	Ladoor SE		2 Jaint (Item 5		L	1 Al	
1111111111111					n 7 item 7		
aution:	Observe a	igebraic sig	ns in combini	ing items!	n 2 Item 3	1tom 4):	
aution:	Observe a	igebraic sign	ns in combin	ing items!	n 2 item 3		
Directivit	Observe a	lgebraic sign	ns in combin	licable, from Tal	n 2 Item 3		
Directivit	Observe a	f Stack Ope	ening, If App	licable, from Tat	n 2 Item 3 ole 34.		1
antative Caution: Directivit	Observe a vy Effect o Indoor SP	gebraic sign f Stack Ope L to This P	ening, If App	licable, from Tat	n 2 Item 3 ole 34.		
antative Saution: Directivit Santative Saution:	Observe al	gebraic sign f Stack Ope PL to This P gebraic sign	ening, If App coint (Item 7 is in combini	licable, from Tat	n 2 Item 3 ole 34.		
antative Saution: Directivit ientative ientative	Observe a y Effect o Indoor SP Observe al	gebraic sign f Stack Opc 2L to This P gebraic sign	ening, If App ening, If App Coint (item 7 is in combini	icable, from Tal	n 2 Item 3 ble 34.		
Directivit Prectivit entative aution: pproxim	Observe al y Effect o Indoor SP Observe al nate Noise iler Data S	gebraic sign f Stack Ope PL to This P gebraic sign Reduction iupplied by	of Muffler Pl Manufacture	ing items!	n 2 Item 3 ole 34. 68) ith Noise Sourc	e, If Applicable,	from Tables 35-3
Proctivit Proctivit entative aution: pproxim om Mut	Observe al y Effect o Indoor SP Observe al hate Noise iler Data S	gebraic sign f Stack Ope PL to This P gebraic sign Reduction iupplied by	of Muffler P Manufacture	<pre>item 1 - item ing items! ilcable, from Tal /pre>	n 2 Item 3 ole 34. 6)	e, If Applicable,	from Tables 35-3
Directivit Cantative Canta	Observe al Py Effect o Indoor SP Observe al nate Noise fler Data S Indoor SP	gebraic sign f Stack Ope L to This P gebraic sign Reduction upplied by L st Neigh	on in combin oning, If App oring, If App orint (Item 7 is in combini of Muffler Pi Manufacture boring Locat	icable, from Tal	n 2 I tem 3 ols 34. (6) (ith Noise Source (	e, If Applicable,	from Tables 35-3
#### Category Classification for Power Plant Noise as Heard in Various Indoor Functional Activity Areas

Category	<ul> <li>Area (and Acoustic Requirements)</li> </ul>	Noise Criterion Designation
1	Bedrooms, sleeping quarters, hospitals,	NC-20
	residences, apartments, hotels, motels, etc. (for	10 NC 20
	sleeping, resting, retaxing):	140-30
2	Auditoriums, theaters, large meeting rooms,	NC-20
	large conference rooms, churches, chapels, etc.	to
	(for very good listening conditions).	NC-30
3	Private offices, small conference rooms,	NC-30
	classrooms, libraries, etc. (for good listening	to
	conditions).	NC-35
4	Large offices, reception areas, retail shops and	NC-35
	stores, cafeterias, restaurants, etc. (for fair	to
	listening conditions).	NC-40
5	Lobbics, laboratory work spaces, drafting and	. NC-40
	engineering rooms, maintenance shops such as	to
	for electrical equipment, etc. (for moderately fair listening conditions).	NC-50
c	Kinhane landeline shore corone machinary	NC-45
Q	Kitchens, launanas, shops, galayes, machinery	to
	minimum accentable speech communication	NC-65
	no risk of hearing damage).	

#### TABLE 2

#### Octave Band Sound Pressure Level (SPL) Values Associated with the Noise Criterion Designations of Table 1

#### SPL (in dB rs 0.0002 microbar) in Octave Frequency Band

Noise Criterion Designation	63 Hz	125 Hz	250 Hz	500 Hz	1,000 Hz	2,000 Hz	4,000 Hz	8,000 Hz
NC-20		40	33	28	22	19	17	16
NC.26	64	44	37	31	27	24	22	21
NC.30	57	48	41	35	31	29	28	27
NC-36	60	52	45	40	36	34	33	32
NC-40	64	56	50	45	41	39	38	37
NC-45	. 67	60	54	49	46	44	43	42
NC-50	71	84	58	54	51	49	48	47
NC-55	74	67	62	58	56	64	53	52
NC-60	77	71	67	63	61	59	58	57
NC-85	80	76	71	68	66	64	63	62

÷....

#### Estimated Octave Band Sound Power Level (PWL) Values for Casing Noise of Unenclosed Gas and Diesel Reciprocating Engines\*

#### Estimated PWL = "Base PWL" (from table below) + Speed Correction + Fuel Correction (in dB re 10<sup>-12</sup> watt)

Continue Rating o	bus f	"Base PWL" in Octave Frequency Band										
Engino, hp	<u>,</u>	63 Mz	125 Hz	250 Hz	500 Hz	1,000 Hz	2,000 Hz	4,000 Hz	8,000 Hz			
15-	23	95	99	99	98	98	97	91	84			
24-	37	97	101	101	100	· 100	99	93	86			
38-	59	99	103	103	102	102	101	95	88			
60-	94	101	105	105	104	104	103	97	90			
95-	149	103	107	107	106	106	105	99	92			
150-	239	105	109	109	108	108	107	101	94			
240-	379	107	111	111	110	110	109	103	96			
380-	599	109	113	113	112	112	111	105	98			
600-	949	111	115	115	114	114	113	107	100			
950-1,	499	113	117	117	116	116	115	109	102			
1,500-2,	399	115	119	119	118	118	117	111	104			
2 40n-3	ROO	117	121	121	120	120	110	112	106			

For Engine Speed:	Speed Correction (in all bands)
Under 600 rpm	—5 dB
600—1,500 rpm	—2 dB
Above 1,500 rpm	0 dB
For Engine Fuel:	Fuel Correction (in all bands)
Natural Gas Only**	—3 dB
Liquid Fuel Only	0 dB
Gas and/or Liquid Fuel	O dB

This table is generally applicable for determining noise control designs of the casing noise of all engines, even though the actual PWL values would not hold for some large angines with unducted turbochargers or unmuffled Roots blowers opening directly into the room.

""With or without a small amount of "pilot oil,"

### Estimated Octave Band Sound Power Level (PWL) Values for Unmuffled Exhaust Noise of Gas and Diesel Reciprocating Engines

# Estimated PWL = "Base PWL" (from table below) + Turbocharger Correction + Exhaust Pipe Length Correction (in dB re 10<sup>-12</sup> watt)

Continu	ous	"Base FWL" in Octave Frequency Band									
Engine, hp		63 Hz	125 Hz	250 Hz	600 Hz	1,000 Hz '	2,000 Hz	4,000 Hz	8,000 Hz		
15	23	122	128	124	116	112	106	96	88		
24-	37	124	130	126	118	114	108	98	90		
38-	69	126	132	128	120	116	110	100	92		
60-	94	128	134	130	122	118	112	102	94		
95-	149	130	136	132	124	120	114	104	96		
150-	239	132	138	134	126	122	116	106	98		
240-	379	134	140	136	128	124	118	108	100		
380-	509	136	142	138	130	126	120	110	102		
600-	949	138	144	140	132	128	122	112	104		
950-1	,499	140	146	142	134	130	124	114	106		
1.500-2	399	142	148	144	136	132	126	116	108		
2,400-3	,800	144	150	146	138	134	128	118	110		

#### For Air Intake to Engine:

#### Turbocharger Correction (in all bands) -6 dB

0 dB

With Turbocharger Without Turbocharger

For Exhaust Pipe Length from Engine: 0-2ft 3-- 6 ft 7–10 ft 11–14 ft

15-18 ft 19-22 ft Lft

į. ŕ

Exhaust Pipe Length Correction (in all bands) 0 dB

1 dB
2 dB
~-3 dB
4 dB
5 dB
_1 /4 dB

- Abras

#### Estimated Octave Band Sound Power Level (PWL) Values for Untreated Turbocharger Noise at Air Inlet Opening of Gas or Diesel Reciprocating Engine

#### Estimated PWL = "Base PWL" (from table below) + Inlet Air Duct Length Correction (in dB re 10<sup>-12</sup> watt)

Continuos	11	"Base FWL" in Octave Frequency Band									
Engina, hp	_	63 Hz	125 <sup>**</sup> Hz	250 Mz	500 Mz	1,000 Hz	2,000 Hz	4,000 Hz	8,000 Hz		
15	23	90	88	88	89	92	93	92	84		
24-	37	<u>91</u>	89	89	90	93	94	93	85		
38	59	92	90	90	91	94	85	94	86		
60-	94	93	<del>9</del> 1	91	92	95	96	95	87		
95- 1	49	94	92	92	93	96	97	96	88		
150- 2	239	95	93	93	94	97	98	97	B9		
240 3	179	96	94	94	95	98	99	98	90		
380- 6	199	97	95	95	96	99	100	99	91		
600- 9	49	98	96	96	97	100	101	100	92		
950-1,4	99	99	97	97	98	101	102	101	93		
1,500-2,3	99	100	98	98	99	102	103	102	94		
2,400-3,8	00	101	99	09	100	103	104	103	95		

For Inlet Air Duct Longth to Engine:	Inlet Air Duct Length Correction (in all bands)				
0—3ft ·	0 dB				
4 9 ft	-1 d8				
10–15 ft	-2 dB				
16–21 ft	-3 dB				
22-27 ft	-4 dB				
28-33 ft	5 dB				
Lft	-L/6 dB				

68

·....

-14	٩В.	LE	0

#### Estimated Octave Band Sound Power Level (PWL) Values for Casing Noise of Unenclosed\* Gas Turbine Engine

PWL (in dB re 10<sup>-12</sup> watt) in Octave Frequency Band

Contraction of the second s

と言語のというないではない

2201 102 102

4.

Continuous Rating of

Engino, kw	63 	125 Hz	250 Hz	500 Hz	1,000 Hz	2,000 Hz	4,000 <u>Hz</u>	8,000 <u>Hz</u>
200- 329	111	113	114	114	114	114	114	114
330- 529	112	114	115	1 15	115	115	115	115
530- 849	113	115	116	118	116	116	116	.116
850-1,299	114	116	117	117	117	117	117	117
1,300-1,999	115	117	- 118	118	118	118	118	11B
2,000-3,299	116	118	119	119	119	119	119	119
3,300-5,000	117	119	120	120	120	120	120	120

•If the entire engine cating is provided with a thermal insulating cover or an enclosing cabinet, the PWL values given here may be reduced by the following amounts (in dB for the octave bands indicated):

Type 1,	Glass fiber or	mineral wool ti	hermal insula	ation with lig	phtweight fo	oil cover over	the insulati	on:				
	2	2	3	3	3	4	5	6				
Тура <b>2.</b>	Glass fiber or steel or 1/2 in	mineral wool ti , thick plaster o	hermal insula cover over th	stion with m e insulation:	inimum 20	gage alumini	ım or 24 gaş	ie				
	5	5	6	6	7	8	9	10				
Тура 3.	Enclosing met no acoustic ab	Enclosing metal cabinet for the entire packaged assembly, with <i>open</i> ventilation holes and with <i>no</i> acoustic absorption lining inside the cabinet;										
	1	1	2	2	2	2	3	3				
Туре 4.	Enclosing met acoustic absor	Enclosing metal cabinet for the entire packaged assembly, with open ventilation holes and with acoustic absorption lining inside the cabinet:										
	4	4	6	6	7	8	8	8				
Туре Б.	Enclosing met cabinet muffle	al cabinet for t d and with aco	he entire pai ustic absorp	ckaged assem tion lining in	bly, with al	li ventilation linet:	holes into t	hø				
	7	8	9	10	11	12	13	14				

69

## Estimated Octave Band Sound Power Level (PWL) Values for Unmuffled Exhaust Noise of Gas Turbine Engine

PWL (in dB to 10"	<sup>2</sup> watt) in Octave	Frequency	/ Band
-------------------	------------------------------	-----------	--------

۰.

Continuous Batima of		PWL (in dB re 10 <sup>-12</sup> watt) in Octave Frequency Band									
Rating of Engine, kw	63 Hz	125 Hz	250 Hz	500 Hz	1,000 Hz	2,000 Hz	4,000 Hz	8,000 Hz			
200- 329	120	122	122	121	119	117	113	107			
330- 529	122	124	124	123	121	119	115	109			
530- 849	124	126	126	125	123	121	117	111			
850-1,299	126	128	128	127	125	123	119	113			
1,300-1,999	128	130	130	129	127	125	121	115			
2.000-3.299	130	132 -	132	131	,129	127	123	117			
3,300-5,000	132	134	134	133	131	129	125	119			

#### TABLE 8 •

### Estimated Octave Band Sound Power Level (PWL) Values for Unmufiled Air Intake Noise of Gas Turbine Engine

. \_

. . . .

. . \_

Continuous Region of		PWL (in dB ro 10.14 watt) in Octave Frequency band									
Engino, kw	63 Mz	125 Hz	250 Hz	500 Hz	1,000 Hz	2,000 Hz	4,000 Hz	8,000 Hz			
200 329	102	103	103	106	112	117	117	114			
330- 629	105	106	106	109	115	120	120	117			
530- 849	108	109	109	112	118	123	123	120			
850-1,299	111	112	112	115	121	126	126	123			
1,300-1,999	114	115	115	118	124	129	129	128			
2,000-3,299	117	118	118	121	127	132	132	129			
3,300-5,000	120	121	121	124	130	135	135	132			

#### Approximate Attenuation of Various Straight Ducts With and Without Lining (May Be Used in Absence of More Complete Analysis of Duct Attenuation by Methods of "ASHRAE Guide")

Values given are in dB attenuation per foot of duct length, for the eight octave bands from 63 Hz to 8,000 Hz. These values assume duct cross dimensions in the region of 3 to 5 feet. These are very rough estimates and should not be used to analyze the duct losses of a ventilation duct layout. If at all possible use the duct analysis methods of the "ASHRAE Guide."

For elevated temperatures, as in gas turbine exhaust ducts, duct dimensions appear shorter in terms of sound wavelengths and therefore usually less effective in attenuating sound, in absence of more thorough analysis at specific exhaust temperature, take only two-thirds the length of the exhaust duct when estimating the attenuation of a hot exhaust duct.

These values are offered only for use in approximating the duct losses of a typical gas turbine engine inlet or exhaust duct before mufflers are added. Do not use these values to design noise control treatments for ducts.

Type 1. For ducting with no internal or external duct lining (dB per foot):

and a final state of a first state of the st

おとなりないたいないではない

Round Duct: Rectangular:	0.10	0.07	0.03	0.02	0.02	0.02	0.02	0.02
Roctangular:	0.20	0.14	0.07	0.05	0.04	0.04	0,04	0.04

Type 2. For ducting with an external thermal duct lining (dB per foot):

Round Duct:	0,15	0.09	0.04	0.02	0.02	0.02	0.02	0.02
Rectangular:	0,30	0.20	0.09	0.05	0.04	0.04	0.04	0.04

Type 3. For ducting with 1-inch-thick internal acoustic absorption duct lining, material and construction to withstand temperature and flow speed (dB per foot):

Round Duct:	0,15	0,16	0.22	0,6	0.8	8.0	0.6	0.4
Roctangular:	0.30	0.25	0.25	0.7	0.8	0.8	0.6	0,4

Type 4. For ducting with 2-inch-thick internal acoustic absorption duct lining, material and construction to withstand temperature and flow speed (dB per foot):

							-	
Round Duct:	0,20	0.20	0,3	0.9	1.1	1.1	0.0	0.6
Rectangular	0.30	0.28	<u>Λ4</u>	n 0	1 1	1 1	0.0	0.0
seeren Beier -	0,00	0,40	V.7	0.0		1 + 1	0.0	<b>U</b> ,U

tin in the second second second second

- The second 
#### Approximate Attenuation of Various Duct Turns With and Without Lining (May be Used in Absence of More Complete Analysis of Duct Attenuation by Methods of "ASHRAE Guide")

Values given are in dB attenuation (or insertion loss) for the sight octave bands from 63 Hz to 8,000 Hz. These values assume duct cross dimension in the region of 3 to 5 feet. These are very rough estimates and should not be used to analyze the duct losses of a ventilation duct layout, if at all possible use the duct analysis methods of the "ASHRAE Guide."

For elevated temperatures, as in gas turbine exhaust ducts, duct dimensions appear shorter in terms of sound wavelengths and therefore usually less effective in attenuating sound. In absence of more thorough analysis at specific exhaust temperature, use only two-thirds the attenuation given below when applying these data to a hot exhaust duct.

These values are offered only for use in approximating the duct losses of a typical gas turbine engine inlet or exhaust duct before mufflers are added. Do not use these values to design noise control treatments for ducts.

If there is more than one turn in the duct configuration, the attenuation of successive turns can be added together only if the turns are located at least 20 feet apart.

Type 1. For ducting with no internal duct lining, 90° rounded turn of round duct or 90° rounded turn of rectangular duct, with ar without turning vanes (dB per turn):

0	1	2	3	3	3	3	3

Type 2. For ducting with no internal duct lining, 90° square turn of rectangular duct, with turning vanes of chord length less than 6 inches (dB per turn):

- 4	5	4	3	3	3	3	

Type 3. For ducting with continuous 1-inch or 2-inch-thick internal acoustic absorption duct lining, 90° rounded turn of rectangular duct, with or without turning vanes (dB per turn):

1

1 2 3 4 5 6 6 6

Type 4. For ducting with continuous 1-inch or 2-inch-thick internal acoustic absorption duct lining, 90° square turn of rectangular duct, with turning varies of chord length less than 6 inches (dB per turn):

1-inch lining:	1	5	6	6	7	8	9	10
2-inch lining:	1	6	7	8	9	10	11	12

### Room Correction Term for Converting PWL of a Source into the Reverberant SPL in a Room

#### SPL = PWL + Room Correction Term

	Room Correction Term Acoustic Treatment of Room								
Volume of Room (cu ft)	Condition	Condition	Condition	Condition					
Under 1,000	0 dB	0 dB	— 3 dB	— 6 dB					
1.000- 2.100	0	- 3	- 6	- 9					
2.200- 4.600	- 2	- 5	- 8	11					
4,700- 9,900	- 4	- 7	-10	-13					
10.000- 21.000	- 6	- 9	-12	-15					
22.000- 46.000	- 8	11	, -14	-17					
47.000-100.000	10	-13	-16	-19					
Over 100,000	-12	15	18	21					

Condition 1: No significant amount of sound absorption material (less than that of Condition 2).

Condition 2: 10-25% of total room area covered with 3/4 in.-1 in, thick sound absorption material,

Condition 3: 26-50% of total room area covered with 3/4 in,-1 in, thick sound absorption material; or 10-30% of total room area covered with 1-1/2 in,-2 in, thick sound absorption material,

Condition 4: Over 50% of total room area covered with 3/4 in,-1 in, thick sound absorption material; or over 30% of total room area covered with 1-1/2 in, thick sound absorption material.

÷.

5.1

A TANK ST. ALIGATION TO A STATE OF A STATE O

12 9 4 9 4 8 4 1 9 . To " 1 Caller of The State

Rules for Adding SPL or PWL Contributions by "dB Addition"

1. For adding two decibel levels together-

When two decibel values differ by:	Add the following amount to the higher value:				
0 or 1 dB	3 dB				
2 or 3 dB	2 dB				
4 to 8 dB	1 dB				
9 dB or more	0 dB				

2. If there are several levels of the same value, they may be added as follows:

No. of equal levels	Add
2	3 dB
3	5 dB
4	6 dB
5	7 dB
6.7	8 dB
8	' 9 dB
9-10	10 dB
N	10 log N dB

 The individual components can be added in any order. The total, using this simplified procedure, will give an answer which is correct to at least 1 dB.

4. When combining the frequency contributions of different sources, add only noise levels from the same octave frequency band.

#### TABLE 13

#### Approximate Transmission Loss (in dB) of Dense\* Poured Concrete or Solid-Core Concrete Block or Masonry

#### (140-150 lb/cu ft density)

Octave		Thickn	ess of Concrete	or Masonry (ir	nches)	
<b>Frequency</b>	4	0	8	10	12	16
Band	<u> </u>	Арр	roximate Surfa	ca Weight (Ib/M	q ft)	
(Mz)	48	72	96	120	144	192
63	32	33	34	35	36	37
125	34	35	36	37	38	39
250	35	36	38	40	41	43
500	37	40	43	45	47	50
1.000	42	46	50	52	54	56
2.000	40	53	56	58	69	61
1.000	65	58	61	63	84	66
3,000	60	63	66	68	69	70

\*For applications involving "transmission loss" as an acoustic requirement, do not use "cinder block" or other lightweight porous block material.

.

e e contra fe

ģ

 $(i,j) \in \{i,j\}$ 

· · · · · · · · ·

Octove

#### Approximate Transmission Loss (in dB) of Hollow-Core Dense\* Concrete Block or Masonry

Octave		Thio	ckness of Conci	rete Block (incl	101)	
Frequency	<b>~ 4</b>	6	8	10	12	16
Band		Арр	roximate Surfa	ce Weight (Ib/s	g ft)	
{Hz}	28	38	44	52	0	76
63	29	30	31	32	32	33
125	32	33	33	34	34	35
250	33	34	35	36	36	37
500	34	35	36	38	39	42
1,000	37	39	41	43	45	48
2,000	42	46	48	50	52	66
4,000	49	62	54	56	58	60
8,000	65	67	59	61	63	65

For applications involving "transmission loss" as an acoustic requirement, do not use "cinder block" or other lightweight percus block meterial.

#### TABLE 15

#### Approximate Transmission Loss (in dB) of Conventional Stud-Type Partitions<sup>1</sup>

Frequency Band (Hz)	Standard Wood Stud Partition <sup>2</sup>	Staggered Wood Stud Partition <sup>3</sup>	improvement with Insulation <sup>4</sup>
63	15	17	1
125	20	22	2
250	26	30	3
500	34	38	4
1.000	40	44	4
2.000	45	47	5
4.000	43	45	5
8,000	45	47	5

Partitions made with 2-1/2 in, to 3-1/2 in, wide steel stude will approximate the values given here for wood-stud construction.

2 2 x 4 wood studs on 16 in, centers, nailed to 2 x 4 wood plates; 5/6 in, thick gypum board nailed on both sides of studs; fill and tape joints and edges.

3 2 x 4 wood studs staggered on 2 x 6 wood plates, alternate studs supporting separate walls of 5/8 in, thick gypsum board; ali-nailed construction, studs for each wall on 16 in. conters; fill and tape joints and edges.

<sup>4</sup> Installation of either (a) 1/2 in, thick glass fiber board or metal spring clips between studs and gypsum board, or (b) min, 1-1/2 in, thick limply supported lightweight insulation in air space between partitions will produce improvement indicated. For staggered partition, use of both types of insulation will produce twice the improvement shown in the table. Add the "improvement values" to the TL of the stud partition to which the insulation has been added.

1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -

m e e

#### Approximate Transmission Loss (in dB) of Filled Metal Panel Partition and Typical Industrial Acoustic Doors

Octave Frequency Band	Filled Motal Proel	Typical Acoustic Doors <sup>2</sup>		
(Hz)	Partition <sup>1</sup>	4 Inch Thick	6 Inch Thick	
63	22	29	35	
125	26	33	37	
250	31	36	39	
500	36	42	46	
1,000	43	47	50	
2,000	48	53	56	
4,000	50	56	61	
8,000	52	59	65	

<sup>1</sup> Constructed of two 16 gegs steel panels filled with 3 in. thickness of 6-8 lb/cu ft glass fiber or rock wool; joints and edges seeled air-tight.

2 Industrial-type acoustic doors typically constructed of sheet steel exterior facings, 1 in, plywood under the sheet steel, densely packed filler of glass fiber or rock wool; heavy farming and hardware; double gasket seels all around door edge. "Studio-type" acoustic doors usually not as thick and heavy, with more elaborate finish datalia.

#### TABLE 17

#### Approximate Transmission Loss (in dB) of Glass\* Walls or Windows (13 lb/sq ft/in, Surface Density)

Octave	Thickness of Glass (inches)				
Frequency	1/8	1/4	1/2	3/4	
Band	ce Weight (Ib/iq ft)				
(Hz)	1-1/2	3	8-1/2	10	
63	5	11	17	20	
125	11	17	23	24	
250	17	23	25	25	
500	23	25	26	27	
1.000	25	26	27	28	
2,000	26	27	28	29	
1,000	27	28	30	33	
3,000	28	-30	36	39	
-					

\*Special laminated safety glass ("Acousta-Pane," Amorada Glass Corporation) containing one or more viscoelastic layers sandwiched between glass panels will yield 3-B dB higher values than given here for single thicknesses of glass; available in opproximately 1/4 in. to 5/8 in, thicknesses.

76

.

#### Approximate Transmission Loss (in dB) of a Few Typical Double-Glass Windows\*

Octavo Frequency Baod	Glass-Air Space-Glass Thicknesses (inches)				
(Hz)	1/4-1/4-1/4	1/4—1·1/2—1/4	1/4-6-1/4		
63	18	10	20		
125	23	23	24		
250	24	25	28		
500	24	27	31		
1,000	26	31	37		
2,000	- 28	34	40		
4,000	30	37	43		
8,000	36	42	46		

\*Thermal insulation double-glass windows typically have 1/4 in. to 1 in. sealed air space between 1/4 in. to 3/8 in. glass panels. For larger air spaces, individual glass panels should be mounted separately in rubber or neoprane gaskets. For large temperature differences across the window, provide desiccant or small ventilation ports in the inner space to eliminate condensation on the cold glass.

#### TABLE 19

#### Approximate Transmission Loss (in dB) of Wood<sup>1</sup> or Plywood (4 lb/sq ft/in. Surface Density)

Octave		Thickness	of Wood or Plywood	l (inches)	
Frequency	1/4	1/2	1	2 <sup>2</sup>	4
Gend	Approximate Surface Weight (lb/sq ft)				
(Hz)	1	2	4	8	16
63	0	4	11	15	18
125	5	10	18	17	19
260	11	15	18	19	20
500	16	17	19	20	28
1,000	18	19	20	26	32
2,000	19	20	26	32	37
4,000	20	26	32	37	` 41
8,000	28	32	37	41	45

<sup>1</sup>Wood construction requires tongue and groove joints, overlapping joints, or sealing of joints against air leakage. For intermediate thicknesses, interpolate between thicknesses given in table.

<sup>2</sup>For 2-In, solid wood doors that are well-gasketed all around, these values of TL may be used,

#### Approximate Transmission Loss (in dB) of Dense\* Plaster (9 lb/sq ft/in. Surface Density)

	Th	ickness of Plaster (in	chos)		
1/2	3/4	1	1-1/2	2	
	Approximate Surface Weight (Ib/ig ft)				
4-1/2	7	9	13	18	
15	18	21	. 24	28	
21	24	26	27	27	
26	27	27	28	28	
27	28	28	29	29	
28	29	29	30	33	
29	- 30	33 .	37	40	
33	37	40	44	47	
40	44	47	50	53	
	1/2 4-1/2 15 21 26 27 28 29 33 40	Th           1/2         3/4           Approxim           4-1/2         7           15         18           21         24           26         27           27         28           28         20           29         30           33         37           40         44	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Thickness of Plastar (inchos) $1/2$ $3/4$ 1 $1/2$ $3/4$ 1Approximate Surface Weight (lb/sq ft) $4-1/2$ 79 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9 $4-1/2$ 9	

If lightweight, non-porous plaster is used, these TL values may be used for equal values of surface weight. These data must not be used for porous or so-called "acoustic plaster."

If plaster is to be used on typical stud wall construction, estimate the total thickness or weight of the plaster and use the TL values given here for that thickness, but increase the TL values where appropriate so that they are not loss than these given in Table 15 for the applicable stud construction.

#### TABLE 21

Sal Mater and sugar

#### Approximate Transmission Loss (in dB) of Sheet Aluminum (14 lb/sq ft/in. Surface Density)

Octova	Thic	kness of Aluminur	n (in.)			
Frequency	1/16	1/8	1/4			
Band	Approxim	Approximate Surface Weight (Ib/sg ft)				
(Hz)	_1	2	3-1/2			
63	1	7	13			
125	7	13	19			
250	13	19	23			
500	19	23	25			
1.000	23	25	26			
2,000	25	26	27			
4,000	26	_ 27	28			
8,000	27	· 28	32			

### Approximate Transmission Loss (in dB) of Sheet Steel (40 lb/sq ft/in. Surface Density)

Octave	Thic	kness of Steel (in	L)
Frequency	1/16	1/8	1/4
Band	Approximat	te Surface Weight	(Ib/sq ft)
(Hz)	2.1/2	5	10
63	9	15	21
125	15	21	27
250	21	27	33
500	27	33	38
1,000	33	38	39
2,000	38	39	39
4,000	39	- 39	37
8,000	39	37	40

1 ÷

#### TABLE 23

### Approximate Wall or Floor Correction Term "C" For Use in the Relationship NR = TL + "C"

Total Surface Area		Acoustic Treatment	of Receiving Room	
Inside Receiving Room Divided by Area of Common Wall or Floor	Condition	Condition 2	Condition 3	Condition
1.4- 2.7	10 dB	-7 dB	-4 dB	-2 dB
2.8- 6.5	- 7	-4	-2	+1
5.6-10	- 4	-2	+1	+3
11 -21	- 2	+1	+3	+4
22 -43	+ 1	+3	+4	+5
44	+ 1	+4	+5	+6

Condition 1: No significant amount of sound absorption material (less than that of Condition 2).

A. . .

Condition 2: 10-25% of total room area covered with 3/4 in,-1 in, thick sound absorption material.

Condition 3: 26-50% of total room area covered with 3/4 in,-1 in, thick sound absorption material; or 10~30% of total room area covered with 1-1/2 in,-2 in, thick sound absorption material.

Condition 4: Over 50% of total room area covered with 3/4 in,-1 in, thick sound absorption material; or over 30% of total room area covered with 1-1/2 in,-2 in, thick sound absorption material.

l Door or Window Area as Parcent of	ll If TL of Door or Window is Loss Than	III Then, Effective TL of Composite Wall Is Less Than TL of
I GTAI YVAII AITA	TL of Wall by	Original Wall by
40%	3 dB	1 dB
	U to	4
	10	7
	20	11
	20	16
20%	3	1
	6	2
	• 10	4
	15	9
	20	13
10%	3	o <sup>*</sup> *
	6	1
	10	3
	15	6
	20	10
<b>5%</b>	3	0
	6	ŏ
	10	1
	15	4
	20	8
2%	3	0
	6	0
	10	Ĭ
	15	2
	20	5
1%	3	0
• 42	6	ŏ
	10	Ō
	15	1
	20	3

#### Approximate Transmission Loss of a Wall Containing Doors or Windows

TABLE 24

**Example:** The TL of a filled metal partition at 125 Hz is 26 dB (Table 16). An 1/8-inch-thick window, accounting for 40% of the over-all partition area, has a TL of +11 dB (Table 17). What is the TL of the composite partition?

Solution: Table 24 indicates that for a 40% window area (Column I) and a TL wall - TL door of 15 dB, [26 - 11 = 15](fourth line of Column II), the TL of the composite wall = 26 - 11 = 15 dB (fourth line of Column III).

ALL AND A DATE OF A D

#### Approximate Transmission Loss (in dB) of Type 1 Floor-Ceiling Combination

#### (See Paragraph 5.8 of text for description of Type 1)

Octava	Thickness of Dense Concrete Slab (in.)					
Frequency	6	8	10	12		
Band		Approximate Surfa	ce Weight (Ib/sq ft)			
<u>(Hz)</u>		96	120	144		
83	33	34	35	36		
125	35	36	37	38		
250	36	38	40	41		
500	40	43	45	47		
1,000	46	50	52	54		
2,000	53 -	58	58	69		
4,000	58	61	63	64		
8,000	63	66	68	69		

à

#### TABLE 26

#### Approximate Transmission Loss (in dB) of Some Type 2 Floor-Ceiling Combinations

#### (See Paragraph 5.8 of text for description of Type 2)

	Thickness of Dense Concrete Slab (in.)							
Octavo	8	8	10	12				
Frequency		Air Space Between Slab and Suspended						
Band	Acoustic Colling (in.)							
(Hz)		18	24	24				
63	36	37	39	40				
125	38	40	42	43				
250	40	43	46	47				
500	45	49	52	54				
1,000	52	57	60	62				
2,000	69	63	66	67				
4,000	84	68	71	72				
8,000	69	73	76	77				

TY LA IT STANL

dia amin'ny solo

8

#### Approximate Transmission Loss (in dB) of Some Type 3 Floor-Ceiling Combinations

#### (See Paragraph 5.8 of text for description of Type 3)

	Thickness of Dense Concrete Stab (in.)						
Octave	6	8	10	12			
<sup>5</sup> mquenc <del>y</del> Band	Air Space Between Slab and Suspended "High TL" Accustic Calling (in )						
<u>Hz)</u>	15	18	24	24			
63	38	40	42	43			
125	41	43	45	46			
250	43	46	49	50			
500	48	52	55	67			
.000	. 55	60	63	85			
.000	62	66	69	70			
.000	67	71	74	75			
,000	72	76	79	80			

#### TABLE 28

Approximate Transmission Loss (in dB) of Some Type 4 Floor-Ceiling Combinations

#### (See Paragraph 5.8 of text for description of Type 4)

	Thickness of Danse Concrete Slab (in.)				
	6	8	10	12	
Octave		Air Space Botween Suspended Plas	Slab and Resiliently ter Ceiling (in.)		
Frequency	18	24	30	30	
Band		Thickness of Dense	Plaster Celling (in.)		
<u>(Hz)</u>	_1_	1	1.1/2	2	
63	41	43	46	48	
125	45	47	50	53	
250	47	50	54	57	
500	52	56	60	64	
1,000	59	84	88	72	
2,000	66	70	74	77	
4,000	71	75	78	82	
8,000	76	80	84	87	

#### Approximate Transmission Loss (in dB) of Some Type 5 Floor-Ceiling Combinations (See Paragraph 5.8 of test for description of Type 5)

For Floating Floor Slab of 3 Inches Thickness Supported Resiliently 2 Inches Above Structure Slab: Add 3 dB to Table 28 Values

For Floating Floor Slab of 4 Inches Thickness Supported Resiliently 2 Inches Above Structure Slab: Add 4 dB to Table 28 Values

For Floating Floor Slab of 5 Inches Thickness Supported Resillently 2 Inches Above Structure Slab: Add 5 dB to Table 28 Values

#### TABLE 30

Outdoor Distance Term for Determining SPL at Any Distance from a Non-Directional Unobstructed Outdoor Sound Source PWL

SPL = PWL - Outdoor Distance Term

#### **Outdoor Distance Term**

Distance (ft)	63, 125, 250 Mz	500 Hz	1,000 Hz	2,000 Hz	4,000 Hz	8,000 Hz
10-11	18	18	18	18	18	18
12-14	20	20	20	20	20	20
15-18	22	22	22	22	22	22
19-23	24	24	24	24	24	24
24-29	26	26	26	26	26	26
30-37	28	28	28	28	28	28
38-47	30	30	30	30	30	30
4859	32	32	32	32	32	32
60-75	34	34	34	34	34	34
7894	36	36	36	36	36	36
95-111	38	38	38	38	38	38
112-140	40	40	40	40	41	42
141-177	42	42	42	42	43	44
178-223	44	44	44	45	48	47
224-281	46	46	46	47	48	50
282-355	48	48	48	49	50	53
356-447	50	60	51	51	63	66
448-563	52	52	53	54	. 56	59
564-711	54	54	55	66	59	63
712-899	56	57	57	58	62	67
900-1,110	68	59	59	61	66	72
1.120-1.400	60	61	62	64	70	78
1.410-1.770	62	63	64	67	75	85
1,780-2,230	64	65	67	70	79	93
2.240-2.810	66	68	70	74	85	102
2.820-3.550	68	70	72	77	92	114
3,5604,470	70	73	76	82	101	128
4,480-5,630	72	76	79	87	111	145
5.640-7.200	74	78	83	93	123	165

83

5.

#### Approximate Noise Reduction of Outside Noise Provided by Typical Exterior Wall Construction with Open and Closed Windows

Octave Frequency Band (Hz)	Walls with Windows Open <sup>1</sup> (dB)	Walls with Air Vents Open <sup>2</sup> (dB)	Walls with Clowd Windows <sup>9</sup> (dB)
83	9	13	19
125	10	14	20
250	11	15	22
500	12	16	24
1,000	13	17	26
2,000	14	18	28
4,000	15	19	30
8,000	16	20	30

1 Any typical wall construction, with open windows covering about 5% of exterior wall area.

2 Any typical well construction, with small open pir vents of about 1% of exterior well area.

3 Any typical wall construction, with closed windows covering about 10-20% of exterior wall area.

#### TABLE 33

#### Approximate Noise Reduction (in dB) Provided by Dense Woods

(Mixed dociduous and evergreen trees; 20-40 feet height, visibility ponetration of 70 to 100 feet)

Octave Frequency	Excess Attenuation		
Band	(in dB per		
(Hz)	100 feet of woods)		
63	1/2		
125	1		
250	1-1/2		
600	2		
1,000	3		
2,000	4		
4,000	4-1/2		
8.000	5		

1 For everyon 10-20 foot height, use one-half the rate given in the table.

2 For sparse woods of 200-300 foot visibility penatration, use one-half the rate given in the table.

TABLE 32

#### Approximate Noise Reduction (in dB) Provided by a Solid Barrier

(Do not go outside of the tabulated range for attenuation in any bands. See text for discussion. Use one-half of attenuation for D greater than 1 mile.)

Source			н					Receiver
A	R					D		B
Batio			Noise i	Reduction i	n Frequency	Band		_
H <sup>2</sup> /R	63	125	250	600	1,000	2,000	4,000	8,000
(ft)	Hz	Hz	Hz	Hz	Hz	<u>Hz</u>	<u>flz</u>	<u> </u>
02-04	0	n	3	6	9	12	15	18
0.5-0.8	ŏ	ž	5	B	11	14	17	20
0.0- 1.2	Ť	4	7	10 .	13	16	19	22
13-19	3	6	9	12	15	18	21	24
20 - 31	6	ā	11	14	17	20	23	24
32-40	7	10	13	16	19	22	24	24
5 _ 1	ĥ	12	15	18	21	24	- 24	24
9 _12	11	14	17	20	23	24	24	24
13 _20	13	18	19	22	24	24	24	24
Over 20	15	18	21	24	24	24	24	24

84

100.1

÷

#### Approximate Directivity Effect (in dB) of a Large Vertical Exhaust Stack Compared to a Non-Directional Source of the Same Power

#### (See paragraph 6.6 of text for discussion)

Approximate Dimensions	Relative SPL for Indicated Angle from Vertical Axis						
and	0°	45°	60°	90 <u>°</u> •	135°.•		
Octave Band (Hz)	(vort.)			(horiz.)			
Over 10x10 ft					<u>`</u> `		
63250 Hz	8	5	2	- 3	- 4		
500-1,000 Hz	9	6	1	- 9	-11		
2,000-8,000 Hz	10	7	-1	-14	-18		
3x3 ft to 10x10 ft							
63-250 Hz	4	2	0	- 2	- 3		
500-1,000 Hz	5	3	1	- 4	- 7		
2,000-8,000 Hz	6	- 4	-2	- 8	12		
Under 3x3 ft							
63-250 Hz	. O	0	0	0	0		
500-1,000 Hz	0	0	· Ó	- 1	- 2		
2,000-8,000 Hz	0	0	ō	- 2	- 4		

\*For air intake openings, values in the 90° and 135° columns can be increased by 50%,

j.

2



Approximate Noise Reduction (in dB) of Typical Reactive Mufflers Used with Reciprocating Engines

Octave		Muffler Series by Relative Size				
Band (Hz)		Small Size	Medium Sizo	Large Sizo		
		"H	igh Pressure-Drop L	ine"		
63		16	20	25		
125		21	25	29		
250		21	24	29		
500		19	22	27		
1,000		17	20	25		
2,000	•.	15	19	24		
4,000		14	' 18	23		
8,000		14	17	23		
		"''	ow Pressure-Drop 1	.ine''		
63		10	15	20		
125		15	20	25		
250		13	18	23		
500		11	16	21		
1,000		10	15	20		
2,000		9	14	19		
4,000		8	13	18		
8,000		8	13	18 •		

Refer to manufacturers' literature for more specific data,

#### TABLE 36

#### Approximate Noise Reduction (in dB) of 8 ft Long, 4 in. Thick Parallel Baffles Separated by Various Width Air Spaces

Octave Frequency	Width of Air Space 4 Inch & Inch 12 Inch 16 Inch						
Band (Hz)	50%	67%	Percent Open Area 75%	80%	86%		
63	3	2	1	t	0		
125	6	5	3	2	2		
250	16	13	8	6	4		
500	32	25	16	13	10		
1,000	56	38	19	16	12		
2,000	48	35	13	11	8		
4,000	40	26	10	B	6		
8,000	20	18	7	6	4		

Refer to manufacturers' literature for more specific data.

TA	BLE	37
----	-----	----

Approximate Noise Reduction (in dB) of 8 ft Long Parallel Baffles of ' Typical Thicknesses and Separations

Octavo		Thickness of	i Baffie (in.)	
Frequency	8	8	12	16
Band		Space (in.)	e (in.)	
(Hz)	8	12	12	10
63	5	4	5	-11
125	11	8	12	-16
250	20	18	20	-23
500	30	27	30	-32
1.000	27	23	24	-25
2.000	22	19	19	-22
4.000	18	14	14	-17
8,000	15	10	10	-14
		<u> </u>	•.	

#### TABLE 38

"Area Factor" ("AF") for Use in Determining the PWL of an Area "A" That Transmits Sound of Level SPL (See paragraph 6.11 of text for discussion)

 $PWL = SPL + (10 \log A - 10)$ = SPL + "AF" (in dB re 10<sup>-12</sup> w)

Aroa "A" (sq. ft.)	"AF" (dB)	Aroa "A" (sq. ft.)	"AF" (dB)
1.0	-10	10	0
1.25	-9	12.5	1
1.6	8	16	2
2.0	-7	20	3
2.5	-6	25	4
3.2	5	32	5
4.0	-4	40	6
5.0	-3	50	7
6.3 '	-2	63	8
8.0	-1	80	9

#### TABLE 39

### Approximate Noise Reduction (in dB) of Various Lengths of Commercial Duct Mufflers

Frequency Band	Muffler Length				
(Hz)	3 ft	5 ft	7 ft		
	"Low	"Low Pressure-Drop Class"			
63	4	8	10		
125	7	12	15		
250	9	14	19		
500	12	16	20		
1,000	15	19	22		
2,000	16	20	24		
4,000	14	18	22		
8,000	9	14	18		
	"High Pressure-Drop Class"				
63	8	11	13		
125	10	14	18		
250	15	23	30		
600	23	32	40		
1,000	30	38	44		
2,000	36	42	48		
4,000	28	36	42		
8,000	23	30	36		

Refer to manufacturers' literature for more specific data,